



Berkeley Economic Review

UC Berkeley's Premier Undergraduate Economics Journal



| Content

Professor Interview

Professor Ted N	Aiguel	07
Interviewd by	Griffin Shufeldt a	nd Dhoha Bareche

Research Papers

Slum Growth in Peruvivan Districts	13
Pulkit Aggarwal	

Did the Clean Air Act Improve Enviromental Justice Disparities?......56 Jared Jageler

Cover

Amy Ai

| Policies

Submission Policy

Format: Please format your submission as a Microsoft word document or Latex file. We would prefer submissions double-spaced, 12-point, Times New Roman font. Please use the Chicago Manual of Style. Please name your file using the following format: "FirstnameLastname-PaperTitle". Length: No maximum length for theses or original research papers. Plagiarism: We maintain a strict zero tolerance policy on plagiarism. According to the University of North Carolina, plagiarism is "the deliberate or reckless representation of another's words, thoughts, or ideas as one's own without attribution in connection with submission of academic work, whether graded or otherwise." For more information about plagiarism-what it is, its consequences, and how to avoid it— please see the UNC's website. Citations: Use Chicago format for in-text citations and your works cited page. Include a separate works cited page at the end of your work. Send to: editor@econreview.berkeley.edu with the subject line "Fall 2021 Journal Submission: [Name], [Paper Title]"

Please direct any questions you may have to the same email address as above. We will reach out after the above deadlines regarding your acceptance status to our journal. Please note that we no longer accept any op-eds, essays, or articles for consideration.

Disclaimers

The views published in this journal are those of the individual authors or speakers and do not necessarily reflect the position or policy of Berkeley Economic Review staff, the UC Berkeley Economics Department and faculty, or the University of California, Berkeley in general.

Copyright Policy

All authors retain copyright over their original work. No part of our journal, whether text or image, may be used for any purpose other than personal use. For permission to reproduce, modify, or copy materials printed in this journal for anything other than personal use, kindly contact the respective authors.

| Staff

Editors in Chief

Douglas Koehler Pearleen Wang

Research and Editorial

Heads Ria Bhandarkar Aishani Banerjee

Editors Zachary Hagen-Smith Divya Vemulapalli Millie Khosla Atman Mohanty Sean O'Connell Kareena Hargunani Kat Xie Anika Srivastava Larry Lin Katherine McKernan Lorene Li Gavin Zhao

Writers Pallavi Murthy Kateri R Mouawad Olivia Doiron Vaidehi Bulusu Natraj Vairavan Aidan Curran Aayush Singh Sara Jane Betts Samuel Quattrociocchi Kunal Kumar Pooja Bapat Jonathan Li Sze Yu Wang Lea Yamashiro

Finance

Michelle Chen

| Staff

Peer Review

Head Owen Yeung

Associates Neel Gupta Ananya Kapur Zan Song Ray Kim Katya Yudina Owen Yeung Prakash Srivastava Nora Gong Matthew Liu Derek Yao Nicolas Sarullo Samuel Gao Arjun Grover Calvin Tang

Layout and Design

Head Karen Li

Associates Hannah Heinrichs Emily Langton Amy Ai Ashley Gong Angela Li

Web Development

Head Daniel Wong

Associates Diana Valenzuela Yashveer Rondla Chris Spencer Janani Sriram

Social

Jennifer Jackson

| Staff

Marketing

Head Caroline Wang

Associates

Lilian Xu Rachel Lee Zoe Cox Jennifer Park Samara Saigal Christopher Lee Swathy Natarajan Ishwari Patankar Aaron Kyle Wang

Outreach

Head Caroline Wang

Associates Jay Dugar Aditya Joshi Savr Kumar Kate Ngo Sarah Dong Pooja Kumaravel Ishaan Shah Christina Owen Neharika Iyer Samarth Bhutani Isabella Tokarz Ruhi Jha Sejal GoelAaron Kyle Wang

Professional Development

Reanna Varsani Tatiana Nikolaeva

| Editor's Note

Dear BER Reader,

In 2016, the Berkeley Economic Review embarked on its inaugural journey, marking the start of a scholarly endeavor dedicated to showcasing top undergraduate economics research. Since our inception, we have remained steadfast in our commitment to publishing innovative and impactful contributions while nurturing the talents of the economists of tomorrow. Today, BER stands united with over 90 dedicated members, receiving hundreds of submissions and captivating tens of thousands of readers from diverse corners of the world.

Amidst a backdrop of dynamic global shifts, encompassing geopolitical challenges, economic uncertainty, and societal transformations, we commend the unwavering dedication of our authors and team. Despite the complexities we face, we find inspiration in the progress made in critical areas such as sustainable energy and strides towards a more inclusive society.

The arduous yet fulfilling task of our Peer Review team has been to identify exceptional undergraduate research that addresses pressing issues of our time. From delving into the complexities of urban slum growth in Peru to examining the ramifications of environmental policies like the Clean Air Act on equity, these papers delve deep into fundamental questions of fairness and justice that persist in our world. Additionally, we revisit our insightful conversation with Ted Miguel featured in our very first issue, tracing the evolution of his research in parallel with BER's journey since its inaugural issue.

With great enthusiasm, we are thrilled to unveil the 12th volume of the Berkeley Economic Review, a testament to the enduring spirit of scholarly inquiry and the pursuit of excellence.

Sincerely,

Pallavi Murthy & Larry Lin Editors-in-Chief Berkeley Economic Review



Ted Miguel

Interviewed by Griffin Shufeldt & Dhoha Bareche **OPENING INFO:** Ted Miguel is the Oxfam Professor of Environmental and Resource Economics at UC Berkeley. He is a founder of CEGA, and his research focuses on development economics, particularly in sub-Saharan Africa. Two Berkeley Economic Review members got the chance to ask him about his background and research earlier this year.

Shufeldt: So to start us off, you were interviewed by us all the way back in 2016, when this journal had just started. In that interview you mentioned that you were always interested in international development and it was just a question of how to pursue that, whether it be law, engineering or political science. Could you talk a little bit about how you landed on economics?

Miguel: Yeah, it's definitely the case that even when I was a teenager, I was really interested in global poverty and equality, international relations, all those kinds of issues. I wasn't sure exactly how to pursue that. When I was in undergrad, I thought I might do something in environmental engineering. I thought I might do something in political science. You mentioned even international law, for a while I was interested in that. I think for me, since I really liked math and kind of statistics and those kinds of technical issues, economics felt like a natural fit because I could work in this area and then also use some of those tools and skills. So I think maybe that's part of what attracts a lot of people to economics is you can use a bunch of math skills, coding skills, and whatnot, and use those technical skills to tackle hard problems in international development. And then I was very lucky because when I was an undergrad, I started working as a research assistant for Michael Kramer, who ended up being my co-author, my advisor in grad school. I was really inspired by the kind of work he was doing, so that nudged me in that direction of doing economics.

Bareche: You've mentioned previously that when you decided to do development economics, it was a relatively new field. How have things changed since you were in undergrad? How do you see the field moving forward in the future?

tique senectus et. Eget duis at tellus at urna condimentum mattis.

Miguel: Development Economics within US universities and in a lot of universities globally was just a small field in terms of the research output, the number of faculty and the number of students, and part of it was due to limited data. Historically there just weren't as many good data sets to use; that limited the research. But the field has changed so much since then and it has grown a ton. It's one of the biggest fields here in Berkeley among our PhD students and there's a lot of interest in the undergrad classes with multiple faculty doing it. So I think the field has grown and it'll continue for years to come because on some level we're way below what steady state development economics should look like. When I started grad school-this was in the nineties-seventy to eighty percent of the world was living in low and middle income countries, but in a department there might be one person doing development economics out of 20 or 30. People's life experiences globally just weren't being represented at all in economics departments, so I think development economics growing is part of that growing representation of more of the world and I think it'll just continue to and maybe there'll be twice as many development economists in 10 years as there are today.

Bareche: Another follow up to that, is there a reason why you decided to study sub-Saharan African development instead of, let's say, Latin American development?

Miguel: I think for me it really was an intellectual decision. In my first year of grad school, I was asking myself exactly that question, what region should I focus on? I had different opportunities to do some research work in South Asia, Latin America or Sub-Saharan Africa. And actually a very influential thing for me is in my first year of graduate school, I was a grad student at Harvard at that time, Jeffrey Sachs was a professor at Harvard, and at the time he had started doing work in Sub-Saharan Africa. He was doing more research, and he gave a talk to first year grad students about his research. And after hearing his talk, I said, okay, I'm gonna do research in Sub-Saharan Africa. It was that simple. Hearing his arguments about [Africa] is the poorest continent, has the worst health problems, has so much political violence, there's so much need to understand what's going on, but there's been so little research on the region comparatively. That combination of real need and very little research was attractive for me intellectually. That summer, after my first year in graduate school, I went to Kenya to work with Michael Kramer. And that was the summer I set up the deworming project and I just had a great experience professionally, personally, and, and then I continued on that line of research.

Shufeldt: Pivoting a little bit to transparency. You wrote an article along with a few others about measuring how transparency has changed over time within economics, whether that be like posting code or instruments. In different subfields, across labor or development or economic theory, there's been like a really positive increase in transparency. One measure

of transparency that has kind of lagged behind the others is pre-registering hypotheses. Why do you think this may be?

Miguel: It's lagged, although even in the last year or two since I wrote the article you're referring to, the numbers have really taken off a lot. So I think it has lagged, but it's changing. Within development economics now, pre-registering studies has become really standard. Papers in other fields and economics less so. So it's kind of something that development economists are doing and some experimental economists, people who do lab experiments, but not really outside those fields. Some of it has to do with the experimental nature of research. There's RCTs and in experimental research and development, pre-registration is a very natural thing to do for experiments because you have to design the experiment in advance, and so it's natural to register it. For some other studies where people are using administrative data, they often don't feel the need to pre-register, even though it would be very useful to do so. So I think that's been the big reason. You talked about why it has been slower? I think that's been the big divide. Researchers doing observational research, non-experimental research, have been slower to adopt the tool, although some are starting to. I'm not sure how much pre-registration will take off outside of experimental work in economics. I think it's still unknown at this point.

Bareche: Economics has a bad rep for taking a technical and narrow minded approach to certain poverty and other social issues. What's your take on this critique? Does modern day economics do a better job at taking an interdisciplinary approach to the problems facing our world today?

Miguel: I think it's kind of the traditional view or traditional critique of economics that it's just narrow and ignores social factors. Maybe one thing you've seen in ECON 172 is how contemporary development economics is actually pretty interdisciplinary and more attention is paid to issues that I think non-economists don't realize economists are working on. Sometimes when I present to non-economists, whether here on campus or more broadly, the kind of characterizations that are made about what economists do or how economists do research are kind of outdated. So I do think economics has changed a lot. I think the fact that a place like CEGA (Center for Effective Global Action) here from its start dedicated itself to do research on international development in an interdisciplinary way and has affiliates from seven or eight different disciplines who are really active in the center kind of speaks to the contemporary research

enterprise being broader than the kind of caricature of what economics is. So I'm proud of that. I'm proud of what we built at Berkeley and again, hopefully in ECON 172 between the historical work and political work and other things, you guys get a sense that development economists are broader than we are sometimes accused of being.

Shufeldt: I think one concern, in economic research in general, is external validity. Especially in development, where a lot of regression discontinuity is used, and we're estimating a lot of the local treatment effects. How can findings in Mozambique for example, be implemented in Nigeria. How can the external validity concerns be addressed?

Miguel: When we're taking maybe a very specific local finding and applying it to a broader setting, maybe somewhere very far away, that's a great point. And, and I think it's something that, not just our research in economics or development or people who use RCTs, but all research is subject to this critique. I mean, qualitative research is subject to that critique. Almost any research, whether it's experimental, observational, quantitative, qualitative economics, sociology is gonna be based on a certain population, and then it's pretty hard to know how those results travel. So I think it's a fundamental research problem. There is some research in what's called meta-analysis or meta research that does find treatment effects estimated in one setting still do have predictive power in other settings. It isn't like there's no external validity. So you were saying, well, what if there's a study in Madagascar or Mozambique and you want to take it to Nigeria? Odds are if there was a big positive effect in Mozambique, there'll still be some sort of positive effect in Nigeria. We don't know for sure, but at least there's some evidence suggesting that. It does speak to the need for more research. You mentioned Nigeria. Nigeria is a country with 200 million people and there's still not enough research on what works and what policies do in Nigeria. So I would say another answer to this is more research in these big important settings. Then we don't have to worry as much about external validity because we'll have that data.

Bareche: What type of research are you working on right now?

Miguel: I'm working in a few different areas. I'll just mention one thing: I'm working in four or five different areas. We've been studying the effects of cash transfers and we've published some work on this, like what are the effects of cash transfers, which have become pretty popular in low and middle income countries in recent years, on the local economy. What we're doing right now is measuring how cash transfer effects spread out over space and studying those effects over time. We're finding some pretty substantial persistence in the benefits of big cash transfer programsnot just for the recipients themselves, but for the local economy. So that's one of the things I'm really excited about because it is really actionable in terms of policy, and at the same time, the economics of it and the theory is really interesting. So it's kind of like the sweet spot for me in terms of my research.

Bareche: Where do you see the field of developmental economics going in the future?

Miguel: I think for me, the ideal project is something that is really intellectually challenging but also important for policy. If I can work in that intersection, then I'm really happy with my research.

We thank Professor Ted Miguel for sharing his research and insight into the future of the field of Development Economics.

Slum Growth in Peruvian Districts

Pulkit Aggarwal¹ Working Paper - February 2022

Abstract

Rapid urbanization in the developing world has brought different challenges from the historical urbanization experience of today's developed countries. One important consequence of "poor-country urbanization" is the growth and persistence of slums – informal housing characterized by low amenities and low standard of living. This paper presents empirical evidence on the relationship between slum growth and urban wage growth in Peru by exploiting variation in employment and wages across industries and space over time. I compute an exports-growth Bartik-style (Shift-Share) instrument to isolate exogenous variation in wage growth across districts in Peru between 2004 and 2019 and use a cross-sectional first-differences model to estimate the elasticity of slum growth to urban wage growth. I estimate an elasticity of -0.94 for growth in slum population (opposite in direction from estimates in the literature) but no effect on growth in slum population as a share of urban population (unlike global trends).

Keywords: slum growth, wage growth, urbanization, bartik instrument

JEL Codes: F63, O18, P25, R21, R31

¹ University of British Columbia, pulkit.aggarwal.098@gmail.com. I am grateful to my advisors, Dr. Victor Couture and Dr. Jamie McCasland, for their advice and technical assistance. I also thank Idaliya Grigoryeva, Pooya Rashidi Ravari, Akash Uppal, and Carla Colina for valuable discussions and support.

| Content

1	Introduction	3
2	Literature Review	6
3	Data	7
3.1	Housing and Household Characteristics Data	7
3.1.1	Defining Urban Districts	8
3.1.2	Defining Slums	9
3.1.3	Wages and Employment	11
3.2	Trade Data	12
4	Empirical Strategy	13
4.1	Bartik Instrument	15
4.1.1	Bartik Instruments in the Literature	16
4.1.2	Applying the Bartik Instrument	17
4.1.3	Computation	18
4.1.4	Instrument Validity	18
4.1.5	Instrument Relevance	23
4.2	Estimating Equations	25
5	Results and Discussion	26
5.1	Growth in Slum Population	26
5.2	Growth in Slum Population for Low-Wage Earners	28
5.3	Growth in Slum Share	29
5.4	Discussion	30
6	Robustness checks	32
7	Conclusion	33
8	A Appendix	38
8.1	A.1 Robustness Check: Trimmed Data	38
8.2	A.2 Robustness Check: Winsorized Data	39
8.3	A.3 Wage Growth Bartik: Computation	40
8.4	A.4 Wage Growth Bartik: Results	41
9	B Supplemental Tables & Figures	43

1 Introduction

Historically, the largest urban cities of the world have been in rich countries. This trend has gradually reversed in the post-war period due to what has been called "poor-country urbanization" (Glaeser, 2014). In 2015, the list of the top 30 cities with more than 10 million inhabitants (or "megacities") was dominated by cities in developing countries and these cities are projected to grow at a faster rate than cities in the rich world between 2015 and 2030 (UN, 2014). However, the urbanization experience in the developing world has been different from the historical urbanization experience of today's rich world (Chauvin et al., 2017; Bryan et al., 2020). For example, city size used to be a strong indicator of living standards but this relationship has broken down over time (Jedwab and Vollrath, 2019).

One important consequence of rapid urbanization has been the growth and persistence of slums – informal housing characterized by crowding, poor quality housing, and low standards of living. Slum residents have poor health and educational outcomes (Galiani et al., 2017), lower levels of public services (Galiani et al., 2017), reduced incentives to invest in housing (Field, 2007; Nakamura, 2017), and greater exposure to crime (Felbab-Brown, 2011). Presence of slums also has societal costs by contributing to traffic congestion (Fernandes, 2011), groundwater pollution due to inadequate sanitation in slums (Nyenje et al., 2013), and overall lower productivity (Cai et al., 2018). In some cases, poor living conditions in slums can lead to non-monetary disutility of living in slums in the form of human capital outcomes that are worse than in rural areas (Marx et al., 2013). This can cause rural residents to forgo higher consumption of urban areas which contributes to spatial misallocation of labour (Lagakos et al., 2018).

Urban poverty may seem preferable to rural poverty, as shown by the revealed preference of migration from rural areas to slums in the hope of the opportunities afforded by proximity to the city. This hypothesis supports the "modernization" theory where slums are a transitory phenomenon in the process of rapid growth which will give way to formal housing as economic growth becomes widespread (Glaeser, 2011). However, evidence supporting this theory is ambiguous at best. Marx et al. (2013) show that mobility out of slums is very low, even intergenerationally and over decades, which is at odds with the view that slums are transitory. Additionally, Gollin et al. (2016) show that urban income growth

in many developing countries come from exports of capital-intensive industries such as natural resources, which increase inequality and do not improve urban living conditions.

Taken together, these facts are more supportive of the hypothesis that slums are urban poverty traps rather than transitory phases of being in the land of opportunity of cities (Marx et al., 2013). This distinction is important from a policy perspective – considering slums as transitory phases justifies the lack of active government intervention. However, acknowledging slums as urban poverty traps provides strong rationale for interventions that reduce barriers to formalization or improve living conditions in slums (commonly known as "slum upgrading policies"). Understanding the factors behind the growth and persistence of slums is then important to both academics and policymakers. This paper investigates the role of one such factor: urban wage growth.

Two trends in global slum growth are noteworthy. First, the absolute number of slum households has been rising, reaching 880 million in 2014 (UN, 2015). A key source of heterogeneity in this trend is that economically dynamic cities (i.e., cities experiencing substantial growth) have experienced higher rates of slum growth (World Bank, 2009). Second, slum population as a share of urban population has been declining, falling from 39% in 2000 to 30% in 2014 (UN, 2015). This paper investigates both these trends in the case of Peru by asking: what is the elasticity of slum growth to urban wage growth in Peru between 2004 and 2019?

To deal with endogeneity concerns of reverse causality and classical measurement error, I use an instrumental variable approach to isolate exogenous variation in wage growth. The main empirical strategy consists of regressing 2004-2019 slum population growth on 2004-2019 urban wage growth instrumented using an exports growth Bartik-style instrument at the district level. To argue for the validity of the constructed Bartik-style instrument in the Peruvian context, I apply the frameworks proposed in Goldsmith-Pinkham et al. (2020) and Borusyak et al. (2020), two recent methodological papers investigating the theoretical robustness of such instruments.

I estimate an elasticity of -0.94 for slum population growth to urban wage growth (significant at the 10% significance level). The magnitude of this elasticity increases to -1.33 when calculated for only low-wage house-holds (significant at the 5% significance level). The direction of these estimates is unexpected because they imply that higher wage growth

in a district causes a decline in the district's slum population, unlike the findings of Alves (2021) who finds a positive elasticity of 1.7 for low-wage workers due to high wage-induced migration but lack of responsive formal housing markets across Brazilian cities. Estimates in the opposite direction suggest potentially elastic formal housing markets in Peru. While I do not consider the housing market in my model and estimate the parameter in only a partial equilibrium model, I discuss my estimates in the context of the interrelated location choice slum growth models presented in Cavalcanti et al. (2019) and Alves (2021) and suggest potential mechanisms. Alternatively, available data (both qualitative and quantitative) suggest that a more likely mechanism specific to Peru is investments into better housing quality by slum residents themselves as a result of wage growth.

This paper contributes to the literature on the determinants of slum growth. My estimates should be understood as part of a general equilibrium model for Peru (as in Alves (2021) for Brazil). The surprising negative elasticity between slum growth and wage growth also raises questions about how the Peruvian labour and housing markets operate differently from those in other developing countries, especially for low-wage earners. In the absence of a formal general equilibrium model for Peru's labour and housing markets, the mechanism of slum decline I argue for – housing investments by slum residents – also points to a potential path out of urban poverty traps.

I conduct two robustness checks. First, I estimate my primary specification using data trimmed at the 5% level on both sides of the distribution for the outcome and independent variables. Second, I winsorize the data the 5% level instead. My estimates are robust to trimming and winsorizing the data for unconditional specifications and the point estimate deviates only a little from the main estimates when controls are added, although the conditional estimates lose statistical significance.

The rest of this paper is organized as follows: Section 2 reviews the literature on slum growth; Section 3 describes the data sources and key variables; Section 4 explains the empirical strategy; Section 5 presents results and discusses mechanisms; Section 6 conducts robustness checks; and Section 7 concludes.

2 Literature Review

Empirical work on structural estimates on the determinants of slum formation and growth is relatively recent in the urban development literature. Cavalcanti et al. (2019) is one of the first papers to empirically estimate the parameters of a general equilibrium model with informal housing and heterogenous agents (in terms of labour productivity and location choice) using data from one slum in Rio de Janerio. The model has two key ideas. The first is that of an opportunity cost of protecting informal land plots in the form of lost labour income. Evidence for the existence of such opportunity costs in the Peruvian context is documented in Field (2007) who shows that a large land titling program between 1996 and 2003 (which increased security of tenure) led to an increase in labour supply of slum residents. The second idea is that per capita income is positively associated with slum growth, which may be due to high incomes pushing up formal housing rents which prevent slum residents from being able to afford formal housing.

Cai et al. (2018) consider a dynamic model with demand for property rights as endogenous and find that the high cost of obtaining property rights (rather than of protecting informal housing) may be a key source of friction stopping slum residents from formalizing as well, although they calibrate the model using simulations, not real-world data. These ideas make it unclear in which direction wage growth would affect slum growth. If wage growth is large enough, it may allow residents to afford the costs associated with formal housing, causing decline in slum population. On the other hand, if wages increase but are not high enough to afford the costs of formalizing, then the higher wages may allow residents to spend more time protecting their informal plots while maintaining the same level of consumption, potentially causing growth in slum population.

More recently, Alves (2021) expands the general equilibrium model to multiple cities by focusing on the interrelated labour and housing markets across cities in Brazil. He decomposes the labour market into low-income and high-income workers and the housing market into informal (i.e., slums) and formal housing, and presents an important descriptive fact – the number of high-income households living in informal housing is negligible. I use this fact to estimate the relationship between slum growth and wage growth separately for low-income workers. For identification, he uses wage Bartik instruments as shocks to local labour markets and migration Bartik instruments as shocks to the housing markets, innovating over past identification strategies of slum growth models.

The current paper should be understood as estimating one parameter of the model presented in Alves (2021). Unlike the general equilibrium model in Alves (2021), I focus on only the labour market and estimate the impact of the dynamics of this market on slum growth, assuming nondynamic housing markets. Alves (2021) estimates an elasticity of 1.7 for low-income workers but a much smaller and statistically insignificant elasticity for high-wage workers. He concludes that wage growth causes slum growth because low-income households are attracted to high wage growth cities, but because formal housing markets in these cities are relatively inelastic to migration, the new migrants end up living in informal housing. Importantly, note that this mechanism is different from the hypotheses in previous models. While previous models point to the role of costs of formalizing borne by slum residents, Alves (2021) shows that another key source of friction is the inelastic supply of formal housing.

3 Data

3.1 Housing and Household Characteristics Data

Data on housing characteristics, employment, and wages come from the Peruvian National Household Survey (Encuesta Nacional de Hogares -ENAHO). ENAHO is a repeated cross-sectional nationally representative survey at the individual- and household-level conducted annually by the Peruvian Statistical Agency (INEI) since 1998. The survey is national in scope, covering all 26 regions of Peru, both urban and rural, and collects data on demographics and living conditions of the population. Relevant to my research question, module 1 of the survey contains detailed questions on housing quality and living conditions, and module 5 collects data on wages and industry of employment. The first year of my data is 2004 (rather than 1998) because INEI adapted a new methodology in 2004 that has been consistent since then for the modules of interest for my analysis. I focus on a 15-year period from 2004 to 2019 because changes in slum levels in response to changes in wage levels would be a medium- to long-term process, necessitating a long study period to identify the relationship robustly. For example, Alves (2021) studies slum

growth in Brazil over a 20-year period using census data from 1990 and 2010. Restricting the study period to 2019 also avoids any bias resulting from the Covid-19 pandemic in 2020.

For all aggregations up to the district-level and district-industry level, I use survey weights (called "expansion factors" in the raw data) provided by ENAHO to make the sample representative of the population. ENAHO's sample design involves stratified sampling using probabilities proportional to size (PPS), meaning these survey weights correspond to the inverse of the final selection probability of each respondent.

3.1.1 Defining Urban Districts

District boundaries include both urban and rural populations. First, at the district-level, I restrict my sample to districts that are at least partially urban in either 2004 or 2019. In other words, my sample consists of only districts that had at least one household surveyed in an urban Primary Sampling Unit (PSU) in 2004 or 2019. ENAHO defines a PSU to be urban if it has a population of 2,000 habitants or more. Of the 412 districts in my final sample, 309 districts were urban in 2004 only and 391 districts were urban in 2019 only. These urban districts correspond to an average urban population of 65,310 in 2004 and 62,933 in 2019. Second, at the individual-level, I restrict the sample to only residents in urban PSUs in these districts and drop incomplete surveys as indicated by a survey status variable in the data.

As slums are an urban concept, entirely rural districts imply a slum population of zero. This would introduce bias in my results because of concentration of data points with a slum population value of zero. However, districts that were entirely rural in 2004 but at least partially urban in 2019 are central to my research question because they represent urbanization of Peru between 2004 and 2019. For these districts, I include rural households in the sample. Conceptually, interpreting change in slums as measuring change in standard of living in districts that were urban in 2019 (rather than the strictly spatial definition of slums measuring urban housing location choice) allows me to include rural households in my sample. Section 3.1.2 discusses this conceptual interpretation in more detail.

3.1.2 Defining Slums

For my outcome variable, change in slum population, I adapt the UN Habitat's definition of slums an urban household is categorized as a slum if it lacks any of the following five amenities: (i) access to safe water, (ii) access to sanitation facilities, (iii) security of tenure, (iv) structural quality and durability of dwelling, and (v) sufficient living area (UN, 2003). Building on the literature which uses the first two amenities only, I adapt the first three amenities to my data and define a household to be a slum household if it (i) lacks access to a local water network connection inside the house, (ii) lacks access to a toilet inside the house, or (ii) does not have a title for the house². I use this definition for two reasons. First, this definition provides a more accurate measure of slums than the one used in the literature (such as in Alves, 2021) which only considers the first two amenities due to data limitations.

Second, the first three criteria are binary in nature and correspond to specific questions in the ENAHO data, but the literature lacks consensus on the thresholds for the last two criteria. I aggregate the number of slum residents to the district-level to arrive at the district-level slum population. Figure 1 maps the outcome variable, change in slum population between 2004 and 2019, for the 412 districts in my sample.

Note that this definition of slums is robust to administrative definitions of slums that many local and national government agencies use to classify



Notes: Darker shades of red indicate slum decline and darker shades of blue indicate slum growth. Source: Own processing of ENAHO data.

² Note that "security of tenure" for rental housing does not translate to having a title for the house but instead having a formal contract with the landlord. However, due to data limitations, I use only having title as a measure of security of tenure.

Rains et al. (2018) shows that administrative definitions can deviate significantly from ground reality and vary even across agencies within a city by comparing administrative demarcations of slums in India to surveys conducted in these areas and manually coding slum clusters in satellite images. However, also note that slums are intuitively a spatial concept that are not accurately captured by the UN Habitat definition. While this definition would have a high true positive rate because it is well documented across contexts that a majority of slum households have no source of private water, no private latrine, and live in an overall unhygienic environment (Marx et al., 2013), we should also expect a non-zero false positive rate due to the inclusion of households outside slums that lack access to safe water, sanitation, and security of tenure. Hence, my results should be interpreted as indicating urban standard of living rather than strictly urban location choice.

3.1.3 Wages and Employment

I compute my explanatory variable, change in annual urban wages at the district-level, by taking the 2004-2019 difference of average annual wages (monetary, in-kind, and profits from self employment) earned by urban respondents in their main occupation. To compute annual wages, I multiply the total income earned in the previous pay period by the frequency of payment. For example, for respondents paid biweekly, I multiply the wages earned in the last two weeks by 26 to arrive at annual wages. Following Alves (2021), I define low-wage workers as those below the 75th percentile of the wage distribution for each year. The 75th percentile corresponds to 15,365 Sols in 2019 and 6,325 Sols in 2004³. Figure 2 plots the cumulative distribution function of wages showing wage growth for low-wage workers between 2004 and 2019 as indicated by the rightward shift of the wage distribution in 2019.

³ $\,$ This roughly equals US \$43,400 in 2019 and US \$17,850 in 2004 (in constant 2010 US \$).



Figure 2: Cumulative Distribution Function of Wages, 2004 and 2019

Source: Own processing of ENAHO data.

ENAHO also maps the industry of employment of the respondent to the 4-digit International System of Industrial Classification (ISIC) Revision 3 level for both 2004 and 2019. I aggregate employment levels up to the district-industry level at the ISIC Rev. 3 2-digit level (rather than the 4-digit level) in order to have enough observations per industry for inclusion in the computation of the Bartik instrument, while also preserving a certain degree of variation. My final sample includes 33 industries at the ISIC Rev 3 2-digit level.

3.2 Trade Data

I use data on Peru's exports from BACI (Base pour l'Analyse du Commerce International). BACI provides data on bilateral trade flows disaggregated by product categories at the 6-digit Harmonized System 2002 (HS-02) level. This dataset is built by CEPII (Centre d'Etudes Prospectives' et d'Informations Internationals) using data reported by countries to UN Comtrade (UN's International Trade Statistics Database). Trade flows included in this dataset are restricted to those whose value exceeds US \$1000. First, I aggregate Peru's export values to all countries up to the HS02 6-digit product level. Second, to merge this industry-level exports data with district-industry level employment data from ENAHO, I use correspondences from World Integrated Trade Solution (WITS) from the World Bank to convert the HS-02 6-digit product codes to ISIC Rev. 3 4-digit industry codes. Third, I aggregate the export values up to the ISIC Rev. 3 2-digit level and merge with the ENAHO data. Finally, I use Peruvian Consumer Price Index (CPI) and exchange rate data from the World Bank to normalize export values, wages, and rents to constant 2010 Peruvian Sols.

Table 1 presents district-level summary statistics for the two years in my sample. Column 3 shows the two-sample t-test difference and significance level. Slum growth is negative while wage growth is positive, suggesting a negative relationship between the two. Table 2 decomposes these variables by slum status and presents individual-level summary statistics for 2004. Wages of slum residents and rents paid by slum residents are significantly lower than non-slum urban residents.

4 Empirical Strategy

I use a cross-sectional first-differences model and instrument for wage growth using an exports growth Bartik-style instrument. The main outcome variable is 2004-2019 log growth in slum population and main explanatory variable is 2004-2019 log growth in average annual urban wages measured in 2010 Sols. Using log transformations for both the outcome and explanatory variables has two advantages. First, it linearizes the distribution of slum growth and wage growth and reduces the implicitly higher weight of outliers, allowing me to estimate the relationship using a linear model. Second, it allows me to interpret the estimated regression coefficient as an elasticity. Figure 3 presents the raw Ordinary Least Squares (OLS) correlation between the two variables.

Estimating the relationship between slum growth and wage growth using an OLS model suffers from at least two sources of statistical endogeneity. First, we can expect potential reverse causality between slum growth and wage growth. Wage growth affects the purchasing power of the population which in turn would affect demand for slum and formal non-slum housing, affecting changes in slum levels. At the same time, depressed wage growth may be a result of living in slums because of lack of investments in human capital formation (for example, in the form of investments in education or health) and being trapped in a poverty trap (Marx et al., 2013). For Peru specifically, Field (2005) and Field (2007) show that giving property rights to slum residents causes an increase in investment

	(1)	(2)	T-test
	2019	2004	Difference
Variable	Mean (std error)	Mean (std error)	(1)-(2)
Urban Population (in 1000s)	40.74 (2.65)	44.49 (3.23)	-3.75
Slum Population (in 1000s)	17.72 (1.09)	21.05 (1.14)	-3.33**
Slum Share	0.52 (0.01)	0.68 (0.02)	-0.16***
Urban Wages (in 1000s of 2010 Sols)	10.28 (0.21)	3.95 (0.12)	6.32***
Urban Rents (in 1000s of 2010 Sols)	2.34 (0.09)	1.41 (0.09)	0.93***
Urban Female share	0.52 (0.00)	0.50 (0.00)	0.02***
Urban Age	34.09 (0.29)	28.15 (0.26)	5.94***
Urban Gini coefficient	0.18 (0.00)	0.16 (0.00)	0.02***
Number of Districts	412	412	

Table 1: District-Level Summary Statistics, 2004 and 2019

.

Notes: Imputed rents are used for non-rental housing (as provided in ENAHO). * p < 0.10, ** p < 0.05, *** p < 0.01

Table 2: Individual-Level Summary Statistics for 2004, Slum and Non- Slum Urban

	(1)	(2)	T-test
	Non-Slum Urban	Slums	Difference
Variable	Mean (std error)	Mean (std error)	(1)-(2)
Wages (in 1000s of 2010 Sols)	6.25 (0.09)	3.43 (0.04)	2.83***
Rents (in 1000s of 2010 Sols)	2.99 (0.02)	0.83 (0.01)	2.16***
Female share	0.52 (0.00)	0.49 (0.00)	0.02***
Age	29.64 (0.13)	25.72 (0.11)	3.92***
Share Employed	0.39 (0.00)	0.33 (0.00)	0.05***

Notes: Imputed rents are used for non-rental housing (as provided in ENAHO). Employment is defined as having positive earnings. * p < 0.10, ** p < 0.05, *** p < 0.01

in residential quality and an increase in labour supply, providing direct evidence of the causal link from slum status to labour-market outcomes. This reverse causality would bias the estimates upwards. Second, data on wages may suffer from classical measurement error as it is comes from self-reported individual-level surveys. This would introduce attenuation bias and bias the estimates towards 0.

To address these endogeneity concerns, I instrument for wage growth using a Bartik-style exports growth instrument and use a first-differences model to estimate the causal effect of wage growth on slum growth. First-differencing controls for any time-invariant district-level characteristics such as proximity to the coast or mountains (which may affect mobility and migration, and hence slum growth), and instrumenting for wage growth isolates plausibly exogenous variation in wage growth.



Figure 3: Log Slum Growth (Y) vs Log Wage Growth (X)

Source: Own processing of ENAHO data.

4.1 Bartik Instrument

Bartik instruments have been used extensively in the urban economics literature (and in spatial economics more generally) to isolate sources of exogenous variation in local labour demand. The idea is to purge the equilibrium wage of local endogenous labour market characteristics and use the exogenous part of labour demand (Baum-Snow and Ferreira, 2015).

4.1.1 Bartik Instruments in the Literature

Most notably in the trade literature, Autor et al. (2013) estimate the effects of Chinese import competition on US local labour-markets by computing a Bartik-style variable. They compute this variable by interacting national growth in Chinese imports by industry with local labour-market employment composition by industry. This measure proxies for the intensity of Chinese import exposure in US commuting zones and is used as the main explanatory variable (rather than an instrument). They exploit cross-sectional variation in commuting zones and the exogeneity in their analysis comes from instrumenting for Chinese imports to US using Chinese imports to other high-income countries.

Mansour et al. (2021) compute a similar import exposure intensity variable to study the differential effects by gender of Chinese imports on local labour outcomes in Peru, but additionally follow the methodological approach proposed in Goldsmith-Pinkham et al. (2020) to argue for exogeneity of the instrument. Specifically, they check that the baseline level of local labour-market industry shares are not correlated with local labour-market characteristics. The idea here is to verify that industry shares that contribute the most to the cross-sectional variation in import exposure are not correlated with other baseline levels of local labour-market characteristics that may be driving labour-market outcomes.

An application of Bartik instruments that is closest to my empirical strategy is Alves (2021) which estimates, among other things, the elasticity of slum growth to wage growth in Brazil between 1990 and 2010. He constructs a Bartik instrument using national wage growth instead of national import intensity to instrument for local labour-market wage growth. Alves (2021) does not argue for the validity of the instrument like Mansour et al. (2021) and instead relies on the extensive use of Bartik instruments in the literature to justify its validity. However, as two recent methodological papers show, some of the canonical applications of Bartik instruments in the literature (including the dataset used by Autor et al. (2013)) do not satisfy the exclusion restriction when examined using theoretically robust tests (Goldsmith-Pinkham et al., 2020; Borusyak et al., 2020).

4.1.2 Applying the Bartik Instrument

Unlike Alves (2021), the version of the Bartik instrument that uses national wage growth does not have strong predictive power in my sample when controls are included. As Section 4.1.4 discusses, my IV strategy requires inclusion of controls because of potential alternate channels that I must control for in order for the exclusion restriction to hold. When controls are included, the first-stage F-stat for the wage growth Bartik instrument falls from 10.4 to 7.6 and the second stage standard errors become significantly larger than the point estimates, suggesting that the estimates are too noisy for any meaningful precise estimation. Computation and results of the wage growth Bartik instrument are presented in Appendix A.3 and A.4.

Instead, I employ a Bartik instrument which uses log growth in national exports by industry rather than log growth in national wages by industry. Intuitively, this Bartik instrument can be thought of as the intensity of export exposure of a district. However, unlike Autor et al. (2013) and Mansour et al. (2021), I do not use the computed Bartik instrument as the explanatory variable. I use it to instrument for district-level wage growth because we should expect the relationship between exports growth and slum growth to be mediated by wage growth - it is unclear how export growth would directly affect slum growth. Conceptually, we should expect districts with higher export exposure intensity to experience higher urban wage growth (Brambila et al. (2017) find that exporting firms pay higher wages to their workers in Peru). The first-stage coefficients confirm this relationship. The coefficients on the Bartik instrument are positive and statistically significant in Table 4. As such, my main empirical strategy should be understood as a combination of the strategies used in Mansour et al. (2021) - which uses an imports exposure intensity Bartik as the explanatory variable - and Alves (2021) - which uses a wage growth Bartik as an instrument. The following subsections discuss the computation, validity, and relevance of the constructed instrument in more detail.

4.1.3 Computation

The exports growth Bartik instrument is computed by interacting the log growth of national exports by industry (the "shocks") with districts' 2004 employment composition by industry (the "shares") and then summing up to the district-level. Mathematically, the district-level Bartik instrument, *Bartikd*, is

$$Bartik_d^{wage} = \sum_{i=1}^{I} \left[(ln \ Wage_{i,-d,2019} - ln \ Wage_{i,-d,2004}) * \frac{N_{i,d,2004}}{N_{d,2004}} \right]$$

where *Exportsi*,2019 are national exports of industry i in 2019, *Exportsi*,2004 are national exports of industry *i* in 2004, *Ni*,*d*,2004 is employment in industry *i* in district *d*, and *Nd*,2004 is the total employment in all industries in district *d*.

4.1.4 Instrument Validity

4.1.4.1 Source of Exogenous Variation

There are two possible sources of exogenous variation in Baritk-style instruments discussed separately in two recent papers. Borusyak et al. (2020) argue that the exclusion restriction for Bartik-style instruments⁴ boils down to orthogonality between the common national shocks and shock-level unobservables. That is, they argue that the same regression coefficients can be obtained from shock-level IV regressions as well, implying that the source of exogenous variation is provided by national shocks, not local industry shares. This approach requires two conditions. First, the variation in the common national shocks must arise from a natural experiment such that they are as-good-as-random. Second, there must be a large number of national shocks for the law of large numbers to kick in at the level of the shock.

On the other hand, Goldsmith-Pinkham et al. (2020) argue that most applications of Bartik-style instruments are essentially a pooled exposure research design where the heterogeneity in baseline level of local

⁴ They use the term Shift-Share Instrumental Variable (SSIV) rather than Bartik instrument but I use the two terms interchangeably here.

labour-market industry shares measures the differential exposure to common shocks. By showing that a 2-SLS estimator using a Bartik instrument is numerically equivalent to a generalized method of moments estimator using each industry share as a separate instrument, they conclude that the exogeneity of the research design depends on the exogeneity of local industry shares, not national shocks.

While both sides agree that the source of exogenous variation must be determined a priori based on the specifics of the context (rather than applying both frameworks), the lack of clarity and consensus on how to establish an a priori argument is illustrated in the fact that both the papers apply their framework to the same paper - Autor et al. (2013). To guide my argument for the Peruvian context, I focus on one criterion provided in both Goldsmith-Pinkham et al. (2020) and Borusyak et al. (2020), which is also implemented by Mansour et al. (2021) in their study of the effects of Chinese imports on local labour-market outcomes in Peru. A priori exogeneity depends on the number of industries that are central to the identification strategy.

If a large number of industries are important for identification, then there is rationale for shocks being exogenous (as argued in Borusyak et al. (2020)). However, if only a few industries are important, then it is likely that local industry shares provide the exogenous variation (as argued in Goldsmith-Pinkham et al. (2020)).

Following Mansour et al. (2021), I visually investigate the heterogeneity in Peruvian exports by industry at the ISIC Rev. 3 2-digit level. Figure 4 presents export values by industry between 2004 and 2019 for the industries in my sample⁵. Industries such as Agriculture & Hunting, Manufacturing of Basic Materials, Manufacturing of Food & Beverages, and Mining of Metal Ores have experienced high export growth between 2004 and 2019 while most of the other industries have experienced relatively low export growth. This indicates that only a few industries provide the variation in national shocks in my sample, suggesting that my identification depends on the exogeneity of the cross-sectional variation in 2004-level of local district industry shares, not the national-level industry export shocks.

⁵ Note that my investigation into the variation in national shocks differs from Mansour et al. (2021) in that they focus on hinese imports to Peru by industry while I focus on global exports of Peru by industry. This distinction means that industries (and the number of industries) that provide the variation in my sample may be different from those in Mansour et al. (2021).

4.1.4.2 Exclusion Restriction

The exclusion restriction requires that the instrument affect the outcome variable only through the endogenous variable. That is, local industry shares should affect slum growth through no channel other than wage growth. One alternate channel in the Peruvian context is the allocation of housing to workers by their employer. If certain industries provide housing as compensation to their workers, then the share of employment in an industry will directly affect housing demand, potentially affecting slum growth. This is likely in Peru because of its long history of conflict between the mining industry and local communities. Deals struck between mining companies and local communities often involve relocation of residents as a compromise⁶. In my sample, only 79

	Agriculture & Hunting	Electricity	Extra-terristrial org	Fishing	Forestry & Logging	Mfg Basic material
s)	Mfg Chemicals	Mig Clothing	Mfg Communication	Mfg Elec Machinery n.e.c.	Mfg Fabricated Metals	Mig Food & Beverage
1 2010 20	Mig Fuel	Mfg Leather	Mig Machinery n.e.c.	Mfg Medicine	Mig Other Transport	Mfg Paper
Exports (thousands of 2010 5018)	Mig Plastics	Mig Textiles	Mig Tohacro	Mig Vehicles	Mfg Wood	Mfg nec
ports (the	Mfg non-metal mineral	Mining coal	Mining metal oces	Oil & Gas Extraction	Other Business	Other Service
	Other mining & quarrying	Publishing	Recreation	2005 2010 2015 2020	2005 2010 2015 2020	2005 2010 2015 2
	2005 2019 2015 2020	2005 2010 2015 2020		ar		

Figure 4: Peruvian Export by Industry between 2004 and 2019

Notes: The trends that look flat are not because of close to zero export growth in these industries but because of the common Y-axis. Source: BACI (UN Comtrade).

districts in 2004 and 33 districts in 2019 have a non-zero share of the population who were provided housing by their workplace, and the mean of this share across districts is negligible: 1.7% of the district urban population in 2004 and 0.4% in 2019. Figure 9 in Appendix B shows the

⁶ See www.reuters.com/article/us-mmg-peru-insight-idUSKBN1E10JG for an example of a deal that relocated rural households to urban towns.

distribution for 2004 and 2019 and Figure 10 shows the relationship between the change in population share provided housing by the workplace and the computed Bartik instrument. The figure shows no correlation between the two, indicating that this is not a significant alternate channel that would invalidate my instrument.

Having established a priori the source of exogenous variation in my Bartik instrument, I now apply the framework proposed in Goldsmith-Pinkham et al. (2020). They provide three tests for the plausibility of the identifying assumption with Bartik instruments. I apply the first test – investigating the correlation of baseline local employment shares of industries with baseline local characteristics that may mediate the relationship between industry shares and slum growth. The underlying exclusion restriction states that the levels of the shares should be exogenous to changes in the error term (i.e., changes in the outcome variable).

The first step is to calculate Rotemberg weights for each industry which indicate the "sensitivity to-misspecification elasticities." These weights tell us which industry shares get more weight in the identification. Higher the Rotemberg weight for an industry, the more sensitive the endogenous variable (wage growth) is to the employment share of that industry. Using these weights, we can identify the top few industries that are most important to the variation and focus the test of the identifying assumption on these industries, rather than all industries. Due to time constraints, I do not compute Rotemberg weights and instead use a more straightforward heuristic to determine which industry shares are contributing most to the cross-sectional variation. For each industry, I calculate the standard deviation of local industry shares across the districts in my sample with the idea that higher the dispersion of industry shares, relatively more the industry contributes to the cross-sectional variation. Figure 5 ranks the industries in terms of this measure.

The second step is to check the correlation of the baseline level of shares of the industries that provide the most variation with baseline level local characteristics. The economic and statistical significance of the correlations indicate the possibility of alternate channels – if the correlations are insignificant, then we can be confident that the exclusion restriction holds. Based on Figure 5, I choose the top ten industries with the highest dispersion. Table 3 presents the correlation of the overall Bartik instrument and the industry shares of the top five industries with local characteristics. Industry correlates for the next five industries are presented in Table 12 in Appendix B.

First note that the R-squared value for the overall Bartik instrument is quite high – 60% of the variation in the computed instrument can be explained by the covariates (Table 3, column 1). Similarly, 59% of the variation in the 2004 industry shares of Agriculture & Hunting can be explained by the covariates (Table 3, column 2). Second, the coefficients for 2004-level of log rents (row 1) are statistically significant for all columns and economically significant for the



Figure 5: Dispersion of Industry Shares across Districts in 2004

Notes: Only industry shares with a standard deviation of greater than 0.01 are shown for better visualization. Source: Own processing of ENAHO data.

overall Bartik instrument (column 1) and shares of Agriculture & Hunting (column 2). A similar pattern holds for 2004-level female share (row 3). These results point to the instrument potentially affecting slum growth through channels other than wage growth. The Bartik instrument that is interpreted as predicting local labour-demand shocks (affecting wages) is also correlated with local labour-supply characteristics (e.g.: female share). Therefore, I control for the baseline level of log rents and female share in my regressions such that the exclusion restriction holds conditional on controls.

	(1)	(2)	(3)	(4)	(5)	(6)
	Bartik	Agri	Retail	Mining	Health	Hotels
	Instrument	& Hunting	Trade	Metal Ores	& Social Work	& Rest.
Urban Rents 2004 (log)	-0.43***	-0.26***	0.043***	0.0035*	0.022***	0.021***
	(0.02)	(0.01)	(0.01)	(0.00)	(0.00)	(0.00)
Urban Age 2004						
(log)	0.071	0.081	0.0051	-0.034	0.0015	-0.0026
	(0.12)	(0.07)	(0.03)	(0.03)	(0.02)	(0.02)
Urban Female share					0.18**	0.010
	(0.32)	· · · · · · · · · · · · · · · · · · ·	()	()	(0.08)	(0.07)
N	412	412	412	412	412	412
R-squared	0.60	0.59	0.19	0.015	0.11	0.11

Table 3: Industry Share Correlates, Bartik Instrument & Top 5 Industries

Notes: Each column reports results of a single regression. Outcome variable is the computed Bartik instrument in column 1 and 2004-level of an industry's share in columns 2-6. Independent variables are 2004-level local district characteristics. District-level regressions, robust standard errors in parentheses.

*p < 0.10, ** p < 0.05, ***p < 0.01

4.1.5 Instrument Relevance

The first-stage of my instrumental variable approach is:

$$ln WageGrowth_d = \alpha_0 + \alpha_1 Bartik_d + X_d + \mu_d$$

where lnWageGrowthd is the log growth in district-level urban wages between 2004 and 2019, Bartikd is the computed exports growth Bartik instrument, and Xd is a vector of district-level controls (discussed in the next subsection). Table 4 presents the first-stage estimates. The F-stat falls as controls are added in columns 2 and 3 but remains above 10 and the coefficients on the Bartik instrument in all three columns are highly significant at the 1% significance level, indicating that the computed exports Bartik is a relevant instrument with high predictive power for local district-level wage growth in my sample.

	(1)	(3)	
	Log Wage growth	Log Wage growth	Log Wage growth
Bartik instrument	0.51***	0.51***	0.39***
	(0.07)	(0.07)	(0.12)
Slum Population 2004			
(log)		0.03	0.03
		(0.03)	(0.03)
Urban Rents 2004 (log)			-0.08
			(0.06)
Jrban Female share 2004	1		-0.46
			(0.61)
Constant	0.73***	0.45	1.27**
	(0.04)	(0.29)	(0.63)
N	412	412	412
F-stat	51.89	26.08	13.92
R-squared	0.17	0.17	0.18

Table 4: First-statge Regression Estimates

Notes: District-level regressions, robust standard errors in parentheses. Dependent variable is 2004-2019 log growth in average annual urban wages. Bartik instrument is calculated by interacting 2004-2019 log growth in national export value by industry with 2004-level local (district) industry shares (at ISIC Revision 3 2-digit level). Controls in column 2 include 2004-level of slum population. Column 3 adds 2004-level of average urban rents and female share of the district as controls. For non-rental housing, self-reported imputed rents are used (from ENA-HO). For districts that were urban only in either 2004 or 2019, rural households are included for the year these districts were not urban. *p < 0.10, ** p < 0.05, ***p < 0.01

4.2 Estimating Equations

The second-stage equation and my primary specification is:

$ln SlumPopulationGrowth_d = \beta_0 + \beta_1 ln \hat{W}ageGrowth_d + X_d + \varepsilon_d$

where *ln SlumPopulationGrowthD* is the log growth in slum population between 2004 and 2019 for *district D*, *ln WageGrowth*[^] *D* is the instrumented log growth in district-level urban wages between 2004 and 2019 for district *D*, and *XD* is a vector of district-level controls. *XD* includes the log of 2004-level of slum population to control for any differential trends in slum growth explained by the baseline level of slum population in a district, log average rents, and female share of urban population to
to control for alternate channels as discussed in Section 4.1.4. $\beta 1$ is the coefficient of interest which gives us the elasticity of slum growth to wage growth. I do not include log 2004level of urban population as a control because of multicollinearity with log 2004-level of slum population (correlation between the two is 0.74). This is reasonable because slum population as a percentage of urban population is high for many districts in my sample. This is visually evident in Figure 7 in Appendix B which shows a large spike for slum population as a share of urban population at the value of one for 2004.

Note that specification (1) includes the entire urban population of districts. However, as shown in Table 2, wages of slum residents are significantly lower than those of urban non-slum residents. It is plausible that due to the aggregation of slum and urban non-slum residents, the estimates of specification (1) are being driven by characteristics of nonslum residents. For my second specification, I decompose the sample into low-wage earners (those earning below the 75th percentile of the wage distribution) and high-wage earners. The resulting equation for the sample restricted to households with at least one low-wage earner is,

$$ln SlumPopulationGrowth_d^{low} = \gamma_0 + \gamma_1 ln \hat{W} ageGrowth_d^{low} + X_d + \varepsilon_d$$
(2)

where *ln SlumPopulationGrowth^{low}D* is the log growth in slum population for individuals in households with at least one low-wage earner between 2004 and 2019 for district d, lnWageGrowth[^] lowd is instrumented log growth in district-level urban wages for low-wage earners between 2004 and 2019 for district d, and the rest of the variables are defined as in the previous specification but for the restricted sample.

I do not include first-differences of these control variables in the regressions because they may be outcomes of wage growth and hence be bad controls. For example, we can expect high urban wage growth to cause high rent growth (through stronger purchasing power and stronger housing demand). However, because urban population growth is a potential confounder in the relationship between slum growth and wage growth, I control for it in my third specification by using change in slum population as a share of urban population as the outcome variable:

$$SlumShareGrowth_{d} = \theta_{0} + \theta_{1}ln \hat{W}ageGrowth_{d} + W_{d} + \varepsilon_{d}$$
(3)

where *SlumShareGrowthD* is the growth in slum population as a share of urban population between 2004 and 2019 for district D, WD is same as *X*D except it omits 2004-level of slum population and includes 2004-level of slum share, and the rest of the variables are defined as in specification (1). Specification (3) lets me investigate whether the global trend of a decline in slum populations as a share of urban population (UN, 2015) is reflected in the Peruvian context.

5 Results and Discussion

5.1 Growth in Slum Population

Table 5 presents the results of my primary specification. Columns 1-3 contain OLS estimates and columns 4-6 contain 2-SLS estimates with each column progressively adding controls. The coefficient on urban wage growth in column 6 implies that a 1% increase in a district's urban wage causes a 0.94% decrease in the district's slum population.

The larger magnitude of 2-SLS estimates than the OLS estimates suggests two possibilities. First, OLS estimates may be attenuated because the wage data suffer from classical measurement error which the 2-SLS estimates get rid of. Second, this is potentially just a mechanical result of using an instrumental variable to predict wage growth because interpretation of OLS and 2-SLS estimates differs in terms of the variation in the independent variable they use. OLS estimates use all the variation in the independent variable to explain the dependent variable, giving us the Average Treatment Effect (ATE). On the other hand, 2-SLS estimates use only the variation in the independent variable that is explained by the instrument, giving us the Local Average Treatment Effect (LATE). That is, to estimate the elasticity, my IV strategy uses only the variation in local wage growth that is explained by exports growth intensity in a district.

	9	Slum Po	pulatior	1		
	(1)	(2)	(3)	(4)	(5)	(6)
	OLS	OLS	OLS	2SLS	2SLS	2SLS
Urban Wage Growth (log)	-0.26**	-0.25***	-0.08	-1.06***	-1.22***	-0.94*
	(0.10)	(0.09)	(0.09)	(0.25)	(0.25)	(0.55)
Slum Population 2004 (log)		-0.44***	- 0.46***		-0.43***	- 0.44***
		(0.05)	(0.05)		(0.05)	(0.05)
Urban Rents 2004 (log)			0.33***			0.12
			(0.06)			(0.15)
Urban Female share 2004	4		-0.02			-0.72
Constant	-0.05	4.11***	(0.88) 1.93***	0.80***	5.08***	(1.18) 4.40^{**}
- o a o o o o o o o o o o o o o o o o o	(0.13)		(0.69)	(0.26)	(0.57)	(1.77)
N	412	412	412	412	412	412
First-stage f-stat				51.89	26.08	13.92
R-squared	0.02	0.19	0.26			0.04

Table 5: OLS and 2-SLS Estimate. Dependent Variable: Log Growth in

Notes: OLS and 2SLS district-level regressions, robust standard errors in parentheses. Dependent variable is 20042019 log growth in slum population. The independent variable of interest is instrumented 2004-2019 log growth in urban wages. Controls in columns 2 and 5 includes 2004-level of slum population. Columns 3 and 6 add 2004-level of average urban rents and female share of the district as controls. For non-rental housing, self-reported imputed rents are used (from ENAHO). For districts that were urban only in either 2004 or 2019, rural households are included for the year these districts were not urban. * p < 0.10, ** p < 0.05, ***p < 0.01

This implies that the 2-SLS standard errors would be larger than the OLS standard errors because of the limited variation in the independent variable being used in 2-SLS, which is the case in Table 5. An important implication of LATE is that the 2-SLS estimates depend on the choice of the instrument because different instruments would explain different parts of the variation in wage growth. This means that the magnitude of the difference between the OLS and 2-SLS estimates may be different if an alternate instrument were used. As such, this makes the ATE and LATE estimates not strictly comparable.

5.2 Growth in Slum Population For Low-Wage Earners

Table 6 presents the estimation results for the sample restricted to households of low-wage earners. The main coefficient on urban wage growth in column 6 implies that a 1% increase in a district's urban wage for lowwage earners causes a 1.33% decrease in the district's low-wage population living in slums. The larger magnitude than the estimate for the entire sample indicates that lowwage individuals are more responsive to the channel of escaping slums than high-wage individuals. Potential channels of slum decline are discussed in Section 5.4.

Table 6: OLS and 2-SLS Estimates. Dependent Variable: Log Growth inSlum Population, sample restricted to Low-Wage Earner Households

	(1)	(2)	(3)	(4)	(5)	(6)
	(1)	(2)		(4)	(3)	(0)
	OLS	OLS	OLS	2SLS	2SLS	2SLS
Urban Wage Growth (log)	-0.39***	-0.40***	-0.12	-1.09***	-1.28***	-1.33**
((0.11)	(0.10)	(0.12)	(0.22)	(0.21)	(0.64)
Slum Population 2004 (log)		-0.42***	-0.45***		-0.43***	-0.43***
		(0.05)	(0.05)		(0.05)	(0.06)
Urban Rents 2004 (log)			0.35***			0.01
			(0.06)			(0.19)
Urban Female share						
2004			0.06			-1.30
			(0.86)			(1.44)
Constant	-0.92***	3.08***	0.71	-0.35**	3.89***	4.52**
	(0.11)	(0.49)	(0.74)	(0.18)	(0.53)	(2.20)
N	393	393	393	393	393	393
First-stage f-stat				84.13	43.22	29.29
R-squared	0.04	0.19	0.26			

Notes: OLS and 2SLS district-level regressions for low-wage earner sample, robust standard errors in parentheses. Dependent variable is 2004-2019 log growth in slum population. The independent variable of interest is instrumented 2004-2019 log growth in urban wages. Controls in

columns 2 and 5 includes 2004-level of slum population. Columns 3 and 6 add 2004-level of average urban rents and female share of the district as controls. For non-rental housing, self-reported imputed rents are used (from ENAHO). For districts that were urban only in either 2004 or 2019, rural households are included for the year these districts were not urban. *p < 0.10, **p < 0.05, ***p < 0.01

5.3 Growth in Slum Share

Table 7 presents estimation results for specification (3) with the outcome variable as growth in slum share. The 2-SLS coefficients on wage growth lose statistical significance when controls are added in columns 5 and 6. The 95% confidence intervals for the point estimates overlap with 0 but are small, suggesting that the effect of wage growth on growth in slum share is statistically indistinguishable from 0 (rather than the point estimates being too noisy).

(1)	(2)	(3)	(4)	(5)	(6)
OLS	OLS	OLS	2SLS	2SLS	2SLS
-0.17***	-0.09***	-0.10***	-0.50***	0.07	-0.03
(0.03)	(0.02)	(0.02)	(0.08)	(0.08)	(0.10)
	-0.58***	-0.72***		-0.68***	-0.73***
	(0.04)	(0.06)		(0.07)	(0.06)
		-0.07***			
					-0.06**
		(0.02)			(0.03)
		-0.16			-0.11
		(0.23)			(0.24)
0.0	2 0.32**	** 0.99***	0.37***	0.23***	0.81***
· · · ·	/ () (0.19)	(0.08)	(0.05)	(0.31)
412	2 412	412	412	412	412
			51.89	26.08	13.92
0.1	3 0.45	0.47		0.35	0.45
	OLS -0.17*** (0.03) 0.03 (0.03) 412	OLS OLS -0.17*** -0.09*** (0.03) (0.02) -0.58*** (0.04) 0.02 0.32** (0.03) (0.03) 412 412	OLS OLS OLS -0.17*** -0.09*** -0.10*** (0.03) (0.02) (0.02) -0.58*** -0.72*** (0.04) (0.06) -0.07*** (0.02) (0.04) (0.06) -0.77*** (0.02) 0.02 0.32*** (0.03) (0.03) (0.03) (0.03) 412 412	OLS OLS OLS 2SLS -0.17*** -0.09*** -0.10*** -0.50*** (0.03) (0.02) (0.02) (0.08) -0.58*** -0.72*** -0.72*** (0.04) (0.06) -0.07*** (0.02) -0.07*** -0.16 (0.02) 0.32*** 0.99*** 0.37*** (0.03) (0.03) (0.19) (0.08) 412 412 51.89	$\begin{array}{c c c c c c c c c c c c c c c c c c c $

Table 7: OLS and 2-SLS Estimates. Dependent Variable: Growth in SlumShare, 2004-2019

Notes: OLS and 2SLS district-level regressions, robust standard errors in parentheses. Dependent variable is 20042019 growth in slum population as a share of urban population. The independent variable of interest is instrumented 2004-2019 log growth in urban wages. Controls in columns 2 and 4 include 2004-level of slum population as a share of urban population. Colums 3 and 5 add 2004-level of average urban rents and female share of the district as controls. For non-rental houses, self-reported imputed rents are used (from ENAHO). For districts that were urban only in either 2004 or 2019, rural households are included for the year these districts were not urban.

* p < 0.10, **p < 0.05, ***p < 0.01

5.4 Discussion

The negative elasticities for slum population growth are surprising given theoretical models of migration that predict migration to urban areas induced by the urban-rural wage gap (Harris and Todaro, 1970) and recent empirical findings that low-wage earners are more responsive to higher urban wages in terms of migrating to urban areas (Alves, 2021; Diamond, 2016). While it is hard to pin down the mechanism causing the decline in slums without a general equilibrium model which includes the housing market, I briefly discuss two channels here. First, I consider the interrelated models presented in Cavalcanti et al. (2019) and Alves (2021) – these relate to the location choice interpretation of slums. Second, I investigate changes in investments in housing quality by slum residents – this relates to the standard of living interpretation of slums discussed in Section 3.1.2.

In the framework proposed by Cavalcanti et al. (2019), a decline in slum population is consistent with high wage growth such that the opportunity cost of protecting informal plots is so high that slum residents are better off living in formal housing. This must require wage growth to be high enough to afford higher rents of formal housing, putting upward pressure formal housing rents. However, Alves (2021) shows for the case of Brazil that formal housing rents are almost four times more responsive to housing demand shocks than informal housing rents. This implies that informal housing supply is more responsive than formal housing supply to housing demand. For the channel in Cavalcanti et al. (2019) to be the primary channel here would require urban wage growth in Peru to be high enough for formal housing to be both, responsive and affordable. This is unlikely given poor urban planning, burdensome bureaucracy, and lack of adequate zoning in urban Peru which hinders supply of low-income formal housing (Bonilla and Barrantes, 2011).

A more likely mechanism of slum decline in the Peruvian context is investments in housing quality by slum residents themselves. Qualitative interviews with residents in slums in Lima, the capital, indicate that residents and the slum community invest substantially in improving their living conditions (ODI, 2015). Data from ENAHO also suggests that investments in housing have increased between 2004 and 2019. Figure 6 shows that the amount of credit received for home improvements or extensions was on average higher in 2019 than in 2004 by low-wage earners (however this is suggestive evidence at best because only 0.02% of low-wage earners in 2004 and 0.04% in 2019 reported having taken a loan for home improvement).

Note that investments in housing quality also indicate that slum residents are aware of low mobility out of slums. This provides evidence against the "modernization" theory of Glaeser (2011) as well because if slums were transitory, residents would have lower incentive to improve their current housing quality. However, note that this channel also suggests that perhaps slums in Peru do not act as poverty traps either. Through housing investments, residents can improve their living conditions and human capital, leading to better labour-market outcomes and potentially entering a virtuous cycle.

Figure 6: Distrubition of Credit for Home Improvement, 2004 and 2019 hange in Slum Population, 2004-2019



Source: Own processing of ENAHO data.

6 Robustness Checks

I conduct two robustness checks. A visual inspection of Figure 3 suggests that a few data points at the extreme ends of the slum growth or wage growth distribution may be driving the results. To check if the results are robust to removal of these outliers, I first run my specifications with slum growth and wage growth trimmed at the 5% and the 95% level. For my second robustness check, I winsorize the variables at the 5% and the 95% level instead of trimming.

The Bartik instrument still has high predictive power for both modified datasets as indicated by the high first-stage F-stats in Tables 8 and 9 in Appendix A. While the second-stage estimates change a little from the main estimation results (in Table 5), note that the estimate for the main specification with controls (Column 6) only deviates a little. It goes from -0.94 for unmodified data to -0.83 for trimmed data and -0.97 for the winsorized data. They lose significance because the standard errors increase by a little which is expected when the sample size falls (as in trimmed data) or the variation in the data is reduced (as in winsorized data). Even though the estimates lose statistical significance, the small deviation in the point estimate and standard errors is suggestive that the relationship between slum growth and wage growth is not entirely due to the extreme ends of the distribution.

7 Conclusion

I compute an exports-growth Bartik instrument to instrument for wage growth and use a first-differences model to estimate the elasticity of slum growth to urban wage growth between 2004 and 2019 in Peru. I estimate a statistically and economically significant negative elasticity of -0.94, implying that urban wage growth causes a decline in slum population in Peruvian districts. These results are opposite in direction from estimates in the literature for other countries, suggesting that further research into the Peruvian context is required, especially using general equilibrium models that study the dynamic relationship between the urban labour and urban housing markets. In the absence of such models, I propose alternate mechanisms specific to Peru such as housing investments by slum residents to improve their living conditions. An important implication of housing investments by slum residents as a result of wage growth is the possibility of slums not acting as poverty traps.

My estimates are limited by the limited number of districts sampled in ENAHO. While Peru has 1,874 districts in total, ENAHO sampled only 1,259 districts in 2019 and 880 districts in 2004. Data on close to all the districts in Peru would improve the precision of these estimates. Another limitation of my analysis is that the definition of slums I use captures standard of living better than it captures urban location choice. Importantly, the definition I use does not capture the social stigma and social exclusion that many slum residents experience from living in slums – this is an inherently location-based effect. As such, it is hard to interpret my estimates to be a result of location choice by slum residents. Future research should consider a different measure of slums that captures location choice better. One such measure is identification of slum clusters by processing satellite images using machine learning (as in Gechter and Tsivanidis (2020)).

Finally, migration in response to wage growth is likely to be a two-step process (rather than a contemporaneous process) where potential migrants in rural areas first learn about wage growth in urban areas and second, decide to migrate in response. Hence, a natural extension to the analysis would be studying the lagged relationship between slum growth (e.g.: 2006-2019) and wage growth (e.g.: 2004-2017).

8 References

1. Alves, G. (2021). Slum Growth in Brazilian Cities. Journal of Urban Economics, page 103327.

2. Autor, D. H., Dorn, D., and Hanson, G. H. (2013). The China syndrome: Local labor market effects of import competition in the United States. American Economic Review, 103(6):2121–2168.

3. Baum-Snow, N. and Ferreira, F. (2015). Causal Inference in Urban and Regional Economics, volume 5. Elsevier B.V., 1 edition.

4. Bonilla, M. and Barrantes, R. W. (2011). Housing Finance in Peru: What is Holding it Back? Inter-American Development Bank, (November).

5. Borusyak, K., Hull, P., and Jaravel, X. (2021). Quasi-Experimental Shift-Share Research Designs. Review of Economic Studies (Forthcoming).

6. Brambilla, I., Depetris Chauvin, N., and Porto, G. (2017). Examining the Export Wage Premium in Developing Countries. Review of International Economics, 25(3):447–475.

7. Bryan, G., Glaeser, E., and Tsivanidis, N. (2020). Cities in the developing world. Annual Review of Economics, 12:273–297.

8. Cai, Y., Selod, H., and Steinbuks, J. (2018). Urbanization and Land Property Rights. Regional Science and Urban Economics, 70(January):246–257.

9. Cavalcanti, T., Da Mata, D., and Santos, M. (2019). On the Determinants of Slum Formation. Economic Journal, 129(621):1971–1991.

10. Chauvin, J. P., Glaeser, E., Ma, Y., and Tobio, K. (2017). What is different about urbanization in rich and poor countries? Cities in Brazil, China, India and the United States. Journal of Urban Economics, 98:17–49.

11. Diamond, R. (2016). The Determinants and Welfare Implications of US Workers' Diverging Location Choices by Skill: 1980-2000. American Economic Review, 106(3):479–524.

12. Felbab-Brown, V. (2011). Bringing the State to the Slum: Confronting Organized Crime and Urban Violence in Latin America. Lessons for Law Enforcement and Policymakers. Brookings Institute, (December).

13. Fernandes, E. (2011). Regularization of Informal Settlements in Latin America. Technical report.

14. Field, E. (2005). Property Rights and Investment in Urban Slums. Journal of the European Economic Association, 3(May 2005):279–290.

15. Field, E. (2007). Entitled to work: Urban property rights and labor supply in Peru. Quarterly Journal of Economics, 122(4):1561–1602.

16. Galiani, S., Gertler, P. J., Undurraga, R., Cooper, R., Mart'inez, S., and Ross, A. (2017). Shelter from the storm: Upgrading housing infrastructure in Latin American slums. Journal of Urban Economics, 98:187–213. 17. Gechter, M. and Tsivanidis, N. (2020). Spatial Spillovers from Urban Renewal: Evidence from the Mumbai Mills Redevelopment. Working Paper.

18. Glaeser, E. L. (2014). A World of Cities: The Causes and Consequences of Urbanization in Poorer Countries. Journal of the European Economic Association, 12(5):1154–1199.

19. Goldsmith-Pinkham, P., Sorkin, I., and Swift, H. (2020). Bartik instruments: What, When, Why, and How. American Economic Review, 110(8):2586–2624.

20. Gollin, D., Jedwab, R., and Vollrath, D. (2016). Urbanization with and without Industrialization. Journal of Economic Growth, 21(1):35–70.

21. Harris, J. R. and Todaro, M. P. (1970). Migration, Unemployment and Developmnent : A TwoSector Analysis. American Economic Review, 60(1):126–142.

22. Imbens, G. W. (2010). Better LATE than nothing: Some comments on Deaton (2009) and Heckman and Urzua (2009). Journal of Economic Literature, 48(2):399–423.

23. Jedwab, R. and Vollrath, D. (2019). The Urban Mortality Transition and Poor-Country Urbanization. American Economic Journal: Macroeconomics, 11(1):223–275.

24. Lagakos, D., Mushfiq Mobarak, A., and Waugh, M. E. (2018). The Welfare Effects of Encouraging Rural-Urban Migration. NBER Working Paper.

25. Mansour, H., Medina, P., and Velasquez, A. (2020). Import Competition and Gender Differences in Labor Reallocation. IZA Discussion Paper, (13608).

26. Marx, B., Stoker, T., and Suri, T. (2013). The Economics of Slums in the Developing World. Journal of Economic Perspectives, 27(4):187–210.

27. Morales, F. and Pierola, M. D. (2020). Venezuelan Migration in Peru: Short-term Adjustments in ' the Labor Market. IDB Working Paper, (August):40.

28. Nakamura, S. (2017). Does Slum Formalisation without Title Provision Stimulate Housing Improvement? A case of slum declaration in Pune, India. Urban Studies, 54(7):1715–1735.

29. Nyenje, P. M., Foppen, J. W., Kulabako, R., Muwanga, A., and Uhlenbrook, S. (2013). Nutrient Pollution in Shallow Aquifers Underlying Pit Latrines and Domestic Solid Waste Dumps in Urban Slums. Journal of Environmental Management, 122:15–24.

30. Overseas Development Institute (2015). On the Path to Progress: Improving living conditions in Peru's Slum Settlements. (June). Rains, E., Krishna, A., and Wibbels, E. (2019). Combining satellite and survey data to study Indian slums: evidence on the range of conditions and implications for urban policy. Environment and Urbanization, 31(1):267–292.

31. UN-Habitat (2003). The Challenge of Slums: Global Report on Human Settlements 2003. Routledge.

32. United Nations (2015). MDG Progress Report. (June).

33. World Bank (2009). Reshaping Economic Geography. Technical report.

8 Appendix

8.1 A.1 Robustness Check: Trimmed Data

Table 8: Estimates for data trimmed at 5%. Dependent Variable: LogGrowth in Slum Population

Table 8: Estimates for data	a trimmed a	it 5%. Depe	endent Var	iable: Log	Growth in	Slum Population
	(1)	(2)	(3)	(4)	(5)	(6)
	OLS	OLS	OLS	2SLS	2SLS	2SLS
Urban Wage Growth (log)	-0.29**	-0.35***	-0.15	-1.31***	-1.61***	-0.83
	(0.11)	(0.11)	(0.11)	(0.32)	(0.35)	(0.74)
Slum Population 2004 (log)		-0.21***	-0.23***		-0.28***	-0.26***
		(0.04)	(0.04)		(0.05)	(0.05)
Urban Rents 2004 (log)			0.26***			0.15
			(0.05)			(0.13)
Urban Female share 200	4		0.85			1.06
			(0.76)			(0.90)
Constant	0.03	2.09***		1.08***	4.11***	1.48
	(0.12)	(0.41)	(0.63)	(0.32)	(0.74)	(1.89)
N	339	339	339	339	339	339
First-stage f-stat				47.56	27.62	14.46
R-squared	0.02	0.08	0.17			0.06

Table 8: Estimates fo	or data trimmed at 5%.	Dependent Variable:	Log Growth in S	lum Population

Notes: OLS and 2SLS district-level regressions, robust standard errors in parentheses. Dependent variable is 20042019 log growth in slum population. The independent variable of interest is instrumented 2004-2019 log growth in urban wages. Log growth in slums and log growth in wages trimmed at the 5th and 95th percentile. Controls in columns 2 and 5 includes 2004-level of slum population. Columns 3 and 6 add 2004-level of average urban rents and female share of the district as controls. For non-rental housing, self-reported imputed rents are used (from ENA-HO). For districts that were urban only in either 2004 or 2019, rural households are included for the year these districts were not urban. *p < 0.10, ** p < 0.05, *** p < 0.01

8.2 A.2 Robustness Check: Winsorized Data

Table 9: Estimates for data winsorized at 5%. Dependent Variable: Log Growth in Slum Population

Table 9: Estimates for dat	a winsorized	i at 5%. De	pendent V	ariable: Lo	g Growth in	n Slum Population
	(1)	(2)	(3)	(4)	(5)	(6)
	OLS	OLS	OLS	2SLS	2SLS	2SLS
Urban Wage Growth (log)	-0.31***	-0.32***	-0.10	-1.19***	-1.35***	-0.97
	(0.10)	(0.09)	(0.10)	(0.26)	(0.25)	(0.59)
Slum Population 2004 (log)		-0.35***	-0.37***		-0.36***	-0.37***
		(0.04)	(0.04)		(0.04)	(0.04)
Urban Rents 2004 (log)			0.30***			0.11
			(0.05)			(0.14)
Urban Female share 200	94		0.34			0.17
			(0.77)			(0.91)
Constant	0.02	3.37***	1.14**	0.93***	4.54***	3.33**
	(0.12)	(0.35)	(0.57)	(0.26)	(0.48)	(1.62)
N	412	412	412	412	412	412
First-stage f-stat				73.28	36.52	18.80
R-squared	0.03	0.17	0.25			0.07

Notes: OLS and 2SLS district-level regressions, robust standard errors in parentheses. Dependent variable is 20042019 log growth in slum population. The independent variable of interest is instrumented 2004-2019 log growth in urban wages. Log growth in slums and log growth in wages trimmed at the 5th and 95th percentile. Controls in columns 2 and 5 includes 2004-level of slum population. Columns 3 and 6 add 2004-level of average urban rents and female share of the district as controls. For non-rental housing, self-reported imputed rents are used (from ENA-HO). For districts that were urban only in either 2004 or 2019, rural households are included for the year these districts were not urban. * p < 0.10, ** p < 0.05, *** p < 0.01

8.3 A.3 Wage Growth Bartik: Computation

The wage-growth Bartik instrument is computed by interacting the log growth of national average wages by industry with districts' 2004 employment composition by industry and then summing up to the district-level. Note that national average wage growth by industry for a district does not include wage growth in that district. This is important for the exogeneity of the instrument because including the district wage growth in the calculation of the national wage growth might lead local labour supply shocks in that district to (non-trivially) affect the instrument, weakening the interpretation of the instrument as a local labour demand shifter. Mathematically, the districtlevel wage growth Bartik instrument, *Bartik-* D^{Wage} , is

$$Bartik_d^{wage} = \sum_{i=1}^{I} \left[(ln \ Wage_{i,-d,2019} - ln \ Wage_{i,-d,2004}) * \frac{N_{i,d,2004}}{N_{d,2004}} \right]$$

where $WageI_{,-d,2019}$ is the average urban wage in industry *i* in all districts except district *d* in 2019, Wagei,-d,2004 is the average urban wage in industry i in all districts except district d in 2004, $N_{i,d,2004}$ is employment in industry *i* in district *d*, and $N_{d,2004}$ is the total employment in all industries in district *d*.

8.4 A.4 wage Growth Bartik: Results

	(1)	(2)	(3)
	Log Wage growth	Log Wage growth	Log Wage growth
Bartik instrument	0.92***	0.94***	0.19
	(0.29)	(0.29)	(0.33)
Slum Population 2004			
(log)		0.02	0.03
		(0.03)	(0.03)
Urban Rents 2004 (log)			-0.19***
			(0.05)
Urban Female share 2004			-0.45
			(0.55)
Constant	0.30	0.05	2.09***
	(0.22)	(0.38)	(0.61)
N	404	404	404
F-stat	10.39	5.35	7.62
R-squared	0.04	0.05	0.11

Table 10: First-stage Regressions. Dependent Variable: LOg Growth in Urban Wages

Notes: District-level regressions, robust standard errors in parentheses. Dependent variable is 2004-2019 log growth in average annual urban wages. Bartik instrument is calculated by interacting 2004-2019 log growth in national wages by industry (excluding wage growth of the district) with 2004-level local (district) industry shares (at ISIC Revision 3 2-digit level). Controls in column 2 include 2004-level of slum population. Column 3 adds 2004-level of average urban rents and female share of the district. For non-rental housing, self-reported imputed rents are used (from ENAHO). For districts that were urban only in either 2004 or 2019, rural households are included for the year these districts were not urban.

* p < 0.10, ** p < 0.05, *** p < 0.01

Table 11: OLS and 2-SLS Regressions.Dependent Variable: LOg Growth in Slum Population

$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Table 11: OLS and 2-SLS	Regressio	ns. Depen	dent Variał	ole: Log G	rowth in Slu	m Population
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(1)	(2)	(3)	(4)	(5)	(6)
		OLS	OLS	OLS	2SLS	2SLS	2SLS
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	e	-0.24*	-0.21*	-0.04	-2.04***	-2.30***	-6.48
		(0.12)	(0.11)	(0.11)	(0.76)	(0.79)	(11.05)
	Slum Population 2004		-0.44***	-0.47***		-0.41***	
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	*						-0.27
Urban Rents 2004 (log) -1.01 (0.06) (2.30) Urban Female share -0.00 2004 -0.00 (0.89) (6.39) Constant -0.06 4.13*** (0.14) (0.50) (0.70) (0.96) N 404 404 404 404 First-stage f-stat 10.39 5.35 7.62			(0.05)	(0.05)		(0.07)	(0.39)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Urban Rents 2004 (log)			0.33***			-1.01
$\begin{array}{cccc} 2004 & & -0.00 & -2.84 \\ & & & & & & & & & & & & & & & & & & $	(8)			(0.06)			(2.30)
$ \begin{array}{c} (0.89) & (6.39) \\ \hline \text{Constant} & -0.06 & 4.13^{***} & 1.96^{***} & 1.77^{**} & 5.94^{***} & 17.09 \\ \hline (0.14) & (0.50) & (0.70) & (0.76) & (0.96) & (26.17) \\ \hline \text{N} & 404 & 404 & 404 & 404 & 404 \\ \hline \text{First-stage f-stat} & & 10.39 & 5.35 & 7.62 \\ \end{array} $	Urban Female share						
Constant -0.06 4.13*** 1.96*** 1.77** 5.94*** 17.09 (0.14) (0.50) (0.70) (0.76) (0.96) (26.17) N 404 404 404 404 404 404 First-stage f-stat 10.39 5.35 7.62	2004			-0.00			-2.84
(0.14) (0.50) (0.70) (0.76) (0.96) (26.17) N 404 404 404 404 404 404 First-stage f-stat 10.39 5.35 7.62				(0.89)			(6.39)
N 404 404 404 404 404 404 404 First-stage f-stat 10.39 5.35 7.62	Constant	-0.06	4.13***	1.96***	1.77**	5.94***	17.09
First-stage f-stat 10.39 5.35 7.62		(0.14)) (0.50)	(0.70)	(0.76)	(0.96)	(26.17)
e		404	404	404		404	
R-squared 0.02 0.19 0.26	First-stage f-stat				10.39	5.35	7.62
1. Squared 0.02 0.17 0.20	R-squared	0.02	0.19	0.26			

Notes: OLS and 2SLS district-level regressions, robust standard errors in parentheses. Dependent variable is 20042019 log growth in slum population. The independent variable of interest is instrumented 2004-2019 log growth in urban wages. Controls in columns 2 and 5 includes 2004-level of slum population. Columns 3 and 6 add 2004-level of average urban rents and female share of the district as controls. For non-rental housing, self-reported imputed rents are used (from ENA-HO). For districts that were urban only in either 2004 or 2019, rural households are included for the year these districts were not urban. * p < 0.10, ** p < 0.05, *** p < 0.01

9 B Supplemental Tables & Figures

Slum population as a share of urban population (solid vertical lines indicate means 2004 2019

Figure 7. Distribution of Slum Share, 2004 and 2019

Source: Own processing of ENAHO data

Table 12: Industry Share Correlates, Next 5 Industries

	(1)	(2)	(3)	(4)	(5)
	Land Transport	Fishing	Construction	Education	
					Business
Urban Rents 2004 (log)	0.021***	-0.00044	0.014***	0.013	***
Urban Age 2004 (log)	0.016**** (0.)	<i>,</i> , , , , , , , , , , , , , , , , , ,	.00) (0.00) (0		
Urban Age 2004 (log) -(0.018 -0.0084 -0	<i>,</i> , , , , , , , , , , , , , , , , , ,			
Urban Age 2004 (log) -(Urban Female share 2004	,	<i>,</i> , , , , , , , , , , , , , , , , , ,			
0 (0,).018 -0.0084 -0 (0.02)	0.011 -0.003	15 -0.0053 (0.	02) (0.01) (0.02) (0.02)
0 (0,	0.018 -0.0084 -0 (0.02) 0.040	-0.033	-0.036	02) (0.01) (0.11**	0.02) (0.02) -0.0058

Each column reports results of a single regression. Outcome variables are 2004-level of an industry's share in columns 1-5. Independent variables are 2004-level local district characteristics. District-level regressions, robust standard errors in parentheses.

* p < 0.10, ** p < 0.05, ***p < 0.01



Figure 8. First-stage: L0g wage Growth (X) vs Bartik Instrument (Z)

Notes: The clustering of data points at 1.6 on the x-axis is because of districts that employ workers only in one industry - Agriculture & Hunting - meaning the share of employment in Agriculture & Hunting for these districts equals 1. Source: Own processing of ENAHO data and BACI (UN Comtrade) data.





Source: Own processing of ENAHO data





Source: Own processing of ENAHO data

Did the Clean Air Act Improve Environmental Justice Disparities?

Jared Jageler¹ Advised by Gabriel E. Lade² April 2022

Abstract

This paper analyzes the differential impacts of the 1990 Clean Air Act Amendments (CAAA) on the racial pollution exposure gap, also known as the Environmental Justice (EJ) gap. Using recently developed, Census tract-level satellite data of PM2.5 pollution, I test whether CAAA non-attainment status and resulting State Implementation Plans decreased pollution in high-percentage Black and Hispanic areas more than in non-high percentage Black and Hispanic tracts. My results confirm that the CAAA reduced overall pollution concentrations in the U.S. and decreased the absolute level of the Environmental Justice gap. A heterogeneity analysis provides evidence that the results are primarily driven by air quality gains in Black communities in California and the Rust Belt.

Keywords: Environmental Justice, Air Pollution, Clean Air Act, State Implementation Plans, Public Economics

¹ Macalester College. Email: jjageler@macalester.edu

² Assistant Professor of Economics, Macalester College. Email: glade@macalester.edu

Contents

1 Introduction	2
2 Background: Air Pollution	2
2.1 Clean Air Act History and Policy	2
2.2 Air Pollution Types and Sources	3
2.3 Air Pollution Health and Economic Impact	3
2.4 Disparate Health Impacts of Pollutions	3
3 Literature Review	4
3.1 Historical Environmental Justice Literature	4
3.2 Economic and Pollution Impacts of the Clean Air Act	4
3.3 Environmental Justice Outcomes of the Clean Air Act	5
3.4 Contribution to the Literature	5
4 Theoretical Framework	6
4.1 Consumers' Willingness-To-Pay for Clean Air	6
4.2 Defining Welfare Beyond Income	6
4.3 Pollutions Havens vs. the Porter Hypothesis	7
5 Data	8
6 Descriptive Statistics	8
6.1 Variable Description	8
6.2 Nationwide net pollution changes	9
6.3 Maps	10
7 Findings	10
7.1 Model	10
7.2 Regression Results	11
7.3 Discussion and Implications	13
8 Heterogeneity Analysis	13
9 Conclusion	14
Appendix	15
Work Cited	15

1 Introduction

Regulating pollution is among the government's most important and controversial undertakings. On the one hand, regulating pollution has immense importance in preventing adverse human health impacts. Regulations are especially important for historically disadvantaged communities, where pollution can interact with other systemic problems to compound harm and create disparities. On the other hand, regulations can disrupt capital investments, jobs, and other economic activities and decision-making. This dichotomy makes the design and analysis of environmental policies of the utmost importance for public welfare.

This paper evaluates one of the most prominent air pollution enforcement mechanisms in the U.S., State Implementation Plans (SIPs), on pollution disparities across racial and demographic characteristics. Under the Clean Air Act (CAA) and its 1990 Amendments (CAAA), the Environmental Protection Agency (EPA) regulates the release of criteria air pollutants, setting maximum permissible pollution limits known as National Ambient Air Quality Standards (NAAQS). While the EPA sets NAAQS and deems whether areas meet those standards, the Agency leaves enforcement to states. When an area is deemed out of compliance, state, tribal, or local governments develop State Implementation Plans (SIPs), which lay out specific plans for bringing the area into compliance with NAAQS (EPA 2021a). The stricter regulation in non-attainment areas versus all other counties provides a convenient and useful source of variation to examine the effects of CAA regulations. Given the staggered implementation of policies over time (attainment status can "turn on and off" every year based on concentrations), there is ripe opportunity for experimental quantitative research methods to examine the impacts of these regulations.



Figure 1 shows that pollution concentrations of Particulate Matter (PM)³ 2.5 have fallen significantly since 1980. Prior research shows SIPs almost certainly played an important role in that progress (McKitrick 2007, Currie et al. 2020). However, gains are not evenly spread. Of particular concern are violations of Environmental Justice (EJ), defined as differential pollution or other environmental hazards faced by marginalized people, which continue to be documented in the popular press and peer-review literature (Shaw and Younes 2021, Cushing et al. 2018, Hernandez-Cortes and Meng 2021). These disparities fluctuate depending on demographics, economics, and policy changes. As EJ concerns increase in importance in the minds of policymakers, it is worth examining what impacts historic Clean Air Act Amendments' SIP rules have had on pollution exposure for disadvantaged communities. To get at this question, I ask whether the Clean Air Act Amendments decreased the quantitative pollution exposure gap between EJ-impacted communities and the rest of the population. I extend previous work, exploring both national trends in policy impacts across minority⁴ and non-minority communities and conducting heterogeneity analysis of individual SIPs to ask where the most progress and EJ reductions have been made. My results confirm that the CAAA reduced pollution concentrations and that the EJ gap has shrunk since 1981. A heterogeneity analysis provides evidence that my results are driven by the significant gains experienced in the Rust Belt, Tri-State area, and Southern California.

2 Background: Air Pollution

2.1 Clean Air Act History and Policy

The Air Quality Act of 1967 was the country's first air pollution control program.⁵ It established the NAAQS, specifying maximum allowable concentrations for six criteria air pollutants. The 1970 CAA and the 1977 CAAA shaped air policy into what we know today. The CAA requires states to develop SIPs that outline specific steps of how they will meet the NAAQS. SIP requirements are much stricter for regions in non-attainment, regions where a criteria pollutant exceeds the allowed NAAQS

4 Throughout this paper I will use the terms "minority," "people/communities of color," and "POC" interchangeable depending on circumstance to refer to Black and Hispanic individuals. Furthermore, while I would prefer to use the term "Latino," or "Latinx," it is standard practice in the literature to use "Hispanic" due to Census Bureau classifications. This nomenclature is not ideal, but I come to this research with the utmost deference to those communities who have experienced years of environmental racism and other forms of hidden oppression.

³ The 10 and 2.5 in PM10 and PM2.5 mean x micrometers or smaller. See Section 2.2 for further definition.

⁵ The regulatory details are summarized from: (Reitze 2004) and (EPA 2020).

concentration (42 U.S.C. 7401). SIPs are designed by a committee of the local regulatory body, elected officials, and representatives of local organizations and must be subsequently approved by the EPA (EPA 2020). The plans are approved or denied solely on emissions reductions criteria ("Revisions to Appendix..." 2018). As a result, neither justice-based concerns about disproportionate or cumulative impacts nor economic burdens are factored into the decision. The 1977 CAAA introduced the first technology-based performance standards on particulates from fossil-fuel fired power plants (Aldy et al. 2022).

Initially, the EPA prohibited the construction of new major polluting stationary sources⁶ for non-attainment regions. However, they later adopted an offset policy where a new facility could be permitted by paying another facility in the same region to reduce emissions permanently (Reitze 2004). The policy has drawn scrutiny from EJ advocates, who fear that local governments permit new facilities to be built in disadvantaged communities, while allowing pollutant concentrations to decrease in affluent areas within the same county.

After the first ten years of the program, over 60 regions with a population of nearly 100 million remained in non-attainment of the 1977 NO2 and Ozone NAAQS (Reitze 2004). In response, the 1990 CAA Amendments divided attainment status by pollutant into classifications like moderate, serious, and severe. The classifications in some cases trigger a SIP policy mechanism like an offset market, inspection/maintenance program, or command and control requirements. As CAA regulations have been revised, technology evolves, and economic conditions change, states may submit revisions to their SIPs subject to approval by the EPA. One such regulatory change was the adoption of two separate standards for PM10, annual and 24-hour (EPA 2021b). The long-term measure protects against chronic respiratory conditions, while the daily measure protects against respiratory irritation and decreased cognitive function (Shehab 2019).

2.2 Air Pollution Types and Sources

The six main categories of criteria air pollutants regulated by the NAAQS are sulfur dioxide (SO2), nitrogen dioxide (NO2), carbon monoxide (CO), particulate matter (PM10 and PM2.5), photochemical oxidants measured as ozone (O3), and lead (EPA 2021c).⁷

⁶ Defined as facilities with 100+ tons pollution/year, i.e. power plants, manufacturing.

⁷ SO2 is a component of sulfur oxides that is primarily generated by power plants and other industrial facilities and causes respiratory issues in humans, haze in the sky, and chemical reactions that form PM. (Continued on the next page)

The main pollutant of interest for this paper is PM2.5, as it is the most studied within the EJ literature due to its adverse health impacts. Many sources contribute to PM, including direct sources such as smokestacks and fires and indirect sources such as power plants, industrial manufacturing, and automobiles. Unlike the other criteria pollutants, which are chemicals, PM is composed of solid particles and airborne liquid droplets. These particles are 30 times smaller in diameter than a human hair and manifest as dust, soot, and smoke that is inhaled and penetrates the lungs posing a health risk at high exposure levels (EPA 2021d).

2.3 Air Pollution Health and Economic Impact

Pollution negatively affects health outcomes in the short- and long-term, which causes economic harm due to resulting respiratory and cognitive ailments such as low birth weight and reduced educational outcomes. Researchers have documented these impacts across several pollutants and settings (Bell et al. 2010, Stingone and McVeigh 2016). For example, ozone harms individuals with asthma and negatively affects worker productivity, even at levels below NAAQS standards (Zivin and Neidell 2012). Similarly, PM2.5 has constrained the Chinese GDP through reduced labor hours and higher medical expenses (Wu et al. 2017). If pollution reduces productivity, as these findings show, abatement can be seen as a form of human capital investment, boosting productivity and growth. Combined with improved health outcomes, these economic impacts provide a strong motivation for the government to pursue pollution control, as they can strengthen local markets (Chakrabarti and Mitra 2005) and the macroeconomy (Leeves and Herbert 2007).

2.4 Disparate Health Impacts of Pollutions

Communities of color are disproportionately susceptible to pollution's harms due to compounding social/economic factors that correlate with race and income. Recent epidemiological literature links long-term PM2.5 exposure to worsened COVID-19 mortality outcomes (Wu et al.

⁷ The sources and environmental effects of NO2 are largely the same as SO2, though it also reacts with other chemicals in the atmosphere to cause Acid Rain.

CO primarily comes from fossil-fuel combustion such as vehicles and home heating and causes oxygen limitation in the body and long-term risk of heart conditions.

Ground level Ozone is the main ingredient in smog. It is formed by reactions with nitrogen oxides and volatile organic compounds (VOCs) after emission from vehicles, refineries, chemical plants, and other industrial facilities. Those with asthma are at highest risk from ozone, typically on hot summer days. It also has adverse ecological effects.

Lead gets in the air primarily through smelters like metals processing operations and waste incinerators. It has significant adverse health impacts, especially in children who experience negative neurological effects. The EPA's regulation of leaded gasoline was a major success, as air lead levels dropped almost 100% over 30 years.

2020). The EJ literature also correlates higher COVID vulnerability to co-dependent social factors like urban/industrialized areas, low household income, low educational attainment, and Black population share (Hooper et al. 2020). This problem extends beyond COVID and lung cancer. Populations can be vulnerable to death from PM exposure alone. Whether you examine direct mortality or correlated illnesses, fine particulate matter is undoubtedly a health crisis. Pope et al. (2009) estimate a 0.61-year reduction in life expectancy for each 10 µg/m³ increase in sustained exposure to PM2.5. Deryugina et al. (2020) find that higher PM exposure correlates with worse health outcomes and lower socioeconomic status. EJ disparities are pronounced in this context, though positive welfare gains in this century demonstrate that pollution abatement has made a difference. With the Black-White life expectancy gap closing by 1.5 years over a 15-year span (Arias et al. 2019), Currie et al. (2020) calculated that 4% of this improvement could be explained by just a 1 µg/ m3 closure of the Black-White pollution gap.

3 Literature Review

3.1 Historical Environmental Justice Literature

The EJ field is interdisciplinary and was spurred by a grassroots social movement. Its first literature was published in the 1980s (Banzhaf et al. 2019). The field's most famous paper was published in 1987 by the Commission for Racial Justice, which documented a correlation between race and proximity to hazardous waste facilities. Since then, scholars have rigorously shown that low-income and people of color are disproportionately exposed to environmental hazards (Evans et al. 2002, Hsiang et al. 2019, Tessum et al. 2021). Disparities are examined as a determinant of interest across numerous socioeconomic variables- including income/ poverty (Hsiang et al. 2020), and age (Gray et al. 2010). The field, closely intersecting with epidemiology, links pollution exposure to a myriad of maladies. Pope (1991) presented the first seminal work, studying PM10 pollution from steel production in the Utah and Salt Lake Valleys and associated increases in respiratory-related hospital admissions.

A common theme across much of the EJ literature is that the location and intensity of polluting facilities depend on local economic and demographic factors (Ringquist 2005; Mohai and Saha 2007). Communities of color are more likely to house polluters (Bullard et al. 2008). Some authors have offered lower political participation as an explanation of this phenomenon, relying on findings from political science that race, income, and health all impact political participation and clout (Michener 2017). In a hypothetical scenario where a new coal plant is opening down the road, who is more likely to comment against it at a city council meeting - a wealthy homeowner or a renter working two jobs? Research suggests the former. Cicatielo et al. (2015) find across 47 countries that wealth is positively linked to conventional political participation. On the environmental side, Hamilton (1995) finds that race, educational attainment, and homeownership predict a community's ability to mobilize against the entry of polluting facilities into an area, raising the expected costs to firms of locating in particular areas. Confirming this point with data, Gray et al. (2010) find that polluting plants in higher voter-turnout-areas face greater regulatory activity. This hypothesis motivates the location-specific costs described in my theory section.

3.2 Economic and Pollution Impacts of the Clean Air Act

State Implementation Plans' efficacy and their responsibility for overall pollution declines are disputed. Reitz (2004) classifies them as a "failure" due to uneven implementation costs, increasing population and manufacturing, and overly optimistic abatement projections. Greenstone (2004) finds that non-attainment status only played a minor role in the impressive 80% drop in sulfur dioxide (SO2) pollution since 1970. However, Greenstone (2002) also pins specific declines in polluting industrial activities to non-attainment status designation: 590,000 jobs, \$37 billion in capital stock, and \$75 billion (1987\$) of output over the first 15 years of the first CAA Amendments. Shapiro and Walker (2018) hypothesize and assert that the large decrease in manufacturing emissions is largely a result of environmental regulation, making pollution more costly, though this is not directly on SIPs. Auffhammer et al. (2009) estimate the effects of non-attainment status on PM10 concentrations at ground-level monitors. They find a treatment effect of -12.5% and that the treatment effect occurred independently of SIP implementation, indicating a regulatory anticipation effect is present. I address this in my model in Section 7.1. In an overarching CAA literature review, Alby et al. (2022) find consistent evidence that pollution declines more rapidly at air monitors in non-attainment counties than those in attainment.

3.3 Environmental Justice Outcomes of the Clean Air Act

While national impacts of the CAA are well-investigated, the EJ literature related to the Act is burgeoning and informs my research. As previously mentioned, firms can trade pollution permits to offset their emissions when in a non-attainment area. These markets are one abatement method that areas can use in their SIPs under the circumstance that they want to permit a new polluting facility to be built. EJ advocates have criticized permit trading programs for potentially allowing pollution to move into minority/low-income areas (Cushing et al. 2016). Shapiro and Walker (2021) analyze the criteria pollutant offset markets legislated through the CAA. They find no substantial effects on pollution movement to communities based on race or income in twelve prominent offset markets. A study of one specific NOx market in the heavily polluted South Coast Air Basin found pollution reductions that do not vary significantly across demographics (Fowlie et al. 2012). Ringquist (2011) studies the sulfur dioxide allowance trading program and finds that communities with high percentages of Black and Hispanic residents experience fewer imports of sulfur dioxide. So, the evidence is inconclusive on the overall efficacy of SIPs, but strongly suggests that new facilities under non-attainment do not harm the EJ gap individually.

In an important nationwide study, Colmer et al. (2020) find that though particulate pollution levels have dropped overall, the most and least polluted areas in 1981 remain so today. They examine non-attainment status as a predictor variable and find that Census tracts in PM2.5 non-attainment before 2016 are associated with an average decrease of 7.09 percentile rank points against all other tracts between 1981 and 2016, i.e., SIPs are associated with declining pollution levels. They also observe a general, though not universal, narrowing of the pollution exposure gap in disadvantaged communities. This is a timid quantification, and my study explores it further. My paper is most closely related to Currie et al. (2020). The authors study differential impacts of the CAA on pollution exposure. The authors find that the Black-White pollution exposure gap has closed since 2000. They attribute 60 percent of the racial convergence in PM2.5 exposure to the CAA. Konisky (2009) can help explain this phenomenon, as following updated federal guidance in the mid-1990s to address EJ concerns, there was evidence of increased CAA enforcement in Black communities and lowered enforcement in Hispanic and poor communities.

3.4 Contribution to the Literature

The EJ literature examines disparities as a determinant of interest across numerous socioeconomic variables- including race (Bullard et al. 2008), income/poverty (Hsiang et al. 2019), education (Colmer et al. 2020), age (Gray et al. 2010), and others (Hausman and Stolper 2020). As explained in Section 6.2, race is the most important dimension of heterogeneity in PM2.5 outcomes. My data analysis shows that racial concentrations are much more predictive of pollution than income or poverty. As such, I focus on disparities across race. Regarding disparities across varying racial/ethnic groups, most authors have studied pollution and health outcomes for Black and Hispanic communities (Konisky 2009; Ringquist 2011; Shapiro and Walker, 2021; Mansur and Sheriff 2021), due to the risk factors recounted in Section 2.4. In states where EJ policy accounts for additional socioeconomic and health factors, vulnerable groups are often referred to as "disadvantaged communities," and studies such as Hernandez-Cortes and Meng (2021) examine changes in pollution on those communities defined in statute. The foundation for my paper, Currie et al. (2020), studies the Black-White pollution gap over time. To build on findings in the literature and bring the conversation in line with previous papers, I approximate the Currie et al. research design, but study the heterogenous effects of the CAAA on the EJ gap for Black and Hispanic Americans.

Another avenue of my contribution is in the data. Researchers have used many methods to analyze air pollution outcomes. These include using facility-level data (Shapiro and Walker 2018) and pollution dispersal models (Hernandez-Cortes and Meng 2021) to approximate where pollution is experienced. Other analyses utilize ground-monitor data from government-run stations (Auffhammer et al. 2009) combined with Census and geographical data (Pope et al. 2002). A recent innovation in the field is satellite remote sensing, which provides broad measurements going back decades. My dataset (explained fully in Section 5) is drawn from Colmer et al. (2020) and Meng et al. (2018), which combines all three of the methods above to create annual, Census Tract level pollution measures.⁸ Additionally, my data contributes to the literature by connecting the findings of treatment effects across PM10 regulations (Auffhammer et al. 2009) and PM2.5 regulations (Currie et al. 2020) on PM2.5 outcomes.

My basic structure of difference-in-differences and event studies mimics the design of Currie et al. (2020), though it diverges in a few features.

⁸ There are numerous advantages to this approach, which are laid out in Section 5.

Currie's policy variables are aggregated by "Commuting Zones" (clustered at the local labor market level) to replicate how the regulations work in practice. Their dependent variable is pollution exposure at the individual level. For simplicity and data limitations, I analyze policy at the county and pollution at the tract level. Furthermore, Currie combines DiD with unconditional quartile regression to estimate counterfactual pollution distributions. They define 19 PM concentration cutoffs using re-centered influence function regressions, then for each, estimate the effect of non-attainment on the probability of moving above the cutoff. This technique is advantageous as it provides stronger backing for causality. However, such an approach is beyond the scope of my paper, though my regression results across multiple specifications are in line with Currie's results. Other methods in the literature include Shapiro and Walker (2021) who use a similar difference-in-differences model to test pollution and permit trading activity across Black and Hispanic communities, but do not find significant results. Colmer et al. (2020) use rank-rank correlations to effectively analyze distributional changes over time, though they are correlated findings and do not robustly control for confounders to prove an EJ causation claim. I utilize their data to attempt to extend those findings. With an "EJ gap" as an outcome of interest, Hernandez-Cortes and Meng (2021) use trend breaks of emissions to examine the impacts of California's Cap and Trade program. As explained in Section 7.1, my model incorporates elements of these studies to attempt to draw a causal picture of the CAAA.

Last, I conduct a novel heterogeneity analysis. The findings illustrate which non-attainment areas drive my average treatment effect and suggest substantial heterogeneity across the U.S. Future work may link this heterogeneity analysis to specific SIP features to better understand the policy mechanisms and local relationships that shape environmental outcomes drive my results.

4 Theoretical Framework

This section summarizes my theoretical framework, documenting economic theories on firm and individual location decisions that may give rise to observed EJ gaps.

4.1 Consumers' Willingness-To-Pay for Clean Air

A central concept in environmental economics is willingness-to-pay (WTP) for clean air. Economists use consumer purchases in a variety of settings, for example, purchases of indoor air purifiers, to estimate individuals' WTP for clean air (Ito and Shuang 2020). Housing is a key market where economists study these issues. When looking at residential housing, more polluted neighborhoods generally have lower property values, ceteris paribus. Economists use heterogeneity in housing characteristics and pollution to estimate WTP for cleaner air based on consumers' locational choices (Bazhaf et al. 2019).

Coase (1960) first hypothesized that firms might locate in poor neighborhoods due to lower potential compensation by the firm to residents (an evaluation of WTP). Following this logic, poorer populations may sort into polluted areas if they prioritize other essentials rather than environmental quality. This process, broadly referred to as Tiebout sorting, links environmental and other social inequalities as a function of wealth and preferences (Banzhaf and Walsh 2013). While observed pollution differences across neighborhoods can cause such exposure gaps to emerge in theory, recent evidence to suggests disadvantaged communities may also have lower WTP for environmental quality due to "hidden" pollution driven by disparities in information about air quality (Hausman and Stolper 2020).

When applying these concepts to PM2.5 pollution in the United States, the data do not support the theory. My statistical analysis in Section 6 demonstrates that the main delineation of PM2.5 pollution inequality in the U.S. is race, not income. It is a finding supported by Currie et al., who statistically show that income differentials explain almost none of the pollution exposure disparities between Black and White populations. While income disparities still persist on many other forms of pollution, additional theory is needed in this research context. To examine outcomes, I draw upon a broader definition of welfare.

4.2 Defining Welfare Beyond Income

In economic theory, social outcomes and expected utility are often modeled as a function of income, with worse outcomes assigned to the poor (Atkinson 1999). However, the Nobel Prize winning economist Amartya Sen pushed the field of Welfare Economics to quantify a multifaceted, humanist approach. One of Sen's notable contributions was to analyze a person's "capability," rather than their utility when assessing their welfare. Capability refers to their freedom of choice and ability to achieve (Sen 1982). For example, a person's ability/freedom to ride a bicycle is not just determined by their income, but their physical health, knowledge, and environmental endowment.⁹ Any of those commodities enhances one's capability to bike. Heavy air pollution, on the other hand, would deteriorate the performance of a child's lungs, their biking capability, and overall welfare. While I do not specifically model capability, it is a useful framework of welfare when examining EJ outcomes.

To explain racial pollution disparities, Sen might assert that, rather than minorities having lower WTP for clean air, we should look to the complex structural factors and endowments determining welfare outcomes beyond income. Key among those factors is a racial wealth gap that exceeds the income inequality gap in the United States (Williams 2017). The racial wealth gap literature diagnoses this problem and frames this paper's link between race and welfare. There are several proposed explanations. Williams (2017) introduces a Wealth Privilege model where wealth is easily transferred down through generations, a process dominated by Whites and furthering economic stratification. Intergenerational wealth transfers are a consensus cause in the literature of the racial wealth gap (McKernan et al. 2014, Darity et al. 2018, Ashman and Neumuller, 2020). In analyzing the impacts of a lack of generational wealth, Herring and Henderson (2016) study "wealth characteristics." They find Black Americans lag behind Whites in ownership of homes, stock, and businesses, as well as receive lower wealth returns to income, education, age, and the previously mentioned assets.

A lack of these wealth characteristics has translated to communities of color being disadvantaged in terms of economics, health, environment, education, housing, and other compounding factors. But why must there be "communities of color" in the first place? Here, the racial wealth gap manifests as residential segregation by race. There are also multiple explanations in the literature for residential segregation. Historically, White government officials and developers used racist policies such as redlining and racial covenants to exclude Black people from their neighborhoods (if not outright intimidation and violence) (Boustan 2013). Recent studies have demonstrated the long-lasting impacts of these policies. Aaron

⁹ Based on a similar hypothetical presented by Sen (1979).

son et al. (2021) show that 1930's redlining maps led in the following decades to racial segregation and wrought reduced homeownership rates, housing values, and credit access. Whipple (2021) finds that homes with racial covenants in the early 20th century were significantly less likely to be foreclosed during the Great Recession and redlining. In the second half of the 20th century, Logan and Parman (2017) show increasing rates of residential sorting were caused by a combination of White flight, urbanization, and deindustrialization, and increased racial sorting at the household level. Then, as explained in the following section, pollution disparities occur as industrial activity moves into segregated neighborhoods.

4.3 Pollutions Havens vs. the Porter Hypothesis

A key question in the EJ literature is how polluters choose their sites. As summarized in Shadbegian and Wolverton (2010), early location theory papers identified natural resource abundance, labor availability, local wages/unionization, market size/proximity, transportation costs, and various production costs as key criteria for firm location based on the principle of profit maximization. More recent literature has turned its attention to EJ questions such as regulation and community welfare. If individuals in disadvantaged communities have a lower WTP for environmental quality (Bazhaf et al. 2019), firms have lower marginal costs from building facilities in those areas. The theory behind this phenomenon dates back to Olson (1965), who connected it to the free-rider dilemma of public goods. He hypothesized it is cheaper for firms to locate in areas where collective action against expected pollution is less likely. A basic economic model considers this as allocatively efficient, as pollution is produced equal to the value society places on it and the government regulates it as an externality. However, governments often do not properly price the externalities of pollution, resulting in a market failure. Furthermore, EJ scholars advocate for policymakers to integrate the cumulative impacts of pollution on marginalized populations into the social cost of pollution. While the Clean Air Act does not consider disparate impacts, it attempts to enforce that social cost of pollution.

"Pollution havens" are an idea that firms locate where environmental regulations are laxer and hypothesizes a negative impact on firms' competitiveness from increased regulations when they are already sited (Dechezleprêtre and Sato 2017). While the theory is typically applied on an inter-country scale, it has relevancy for intra-country analysis. The hypothesis that environmental regulations, all else being equal, will reduce a firm's economic competitiveness was first raised by McGuire (1982). Pollution havens have been rigorously debated, and their empirical evidence is far from conclusive due to issues of endogeneity (Levinson and Taylor 2008). There is, however, a competing hypothesis, The Porter Hypothesis, that implementing a stringent policy on a firm will trigger the economic incentive for them to invest in pollution abatement technologies. The authors hypothesize further that adopting clean technologies can raise productivity and costs savings to the point where the firm profits from the change in the long run (Porter and van der Linde 1995). This theory and academic conversation provide a narrative of the possible causal relationship around the closing EJ gap. If the Porter Hypothesis holds, communities of color experienced disproportionate gains in clean technology (and possibly industrial productivity). My data exploration and regression results provide quantitative backing for this story, while the heterogeneity analysis introduces additional possibilities.

5 Data

This paper investigates whether the Clean Air Act Amendments decreased pollution exposure disparities, and if so, to what extent were SIPs and non-attainment status responsible for the change? To answer this question, it is necessary to have historical measurements of air pollution that are trustworthy and precise enough to mitigate concerns of environmental fallacy.

Much of the publicly available pollution data in the United States comes from the EPA's monitoring stations. While they are the foundation for much of the country's air quality enforcement, the network is notoriously flawed. First, there are not enough monitors to ensure every American can know their exposure. Not only does monitoring vary by race and income in dense areas (Stuart 2012), there are large swaths of rural America that are not covered at all (Garcia et al. 2016). Perhaps the most significant issue in air monitoring, however, is intermittency. Due to resource limitations, some monitors only collect data once every 6-12 days. With this schedule being public knowledge, some polluters have strategically adjusted their emissions to stay within attainment on those days (Zou 2021).

Herein lies the importance of a recent innovation: satellite data on pol

lution. Zou (2021) uses such data to show that air quality is significantly worse on unmonitored days due to arbitrary measurement techniques. The most relevant study using these methods is Meng et al. (2019), who estimated ambient fine particulate matter concentrations across North America from 1981–2016 through chemical transport modeling, satellite remote sensing, and ground-based measurements. This provides a much more accurate picture of PM2.5 distribution than previously available to researchers, which has since strengthened the corollary claims able to be made.

I use pollution data from Colmer et al. (2020). The authors use satellite data of PM concentrations across the U.S. from Meng et al. to construct annual PM measures at the Census tract level. The authors construct the measures using satellite data, a chemical transport model, and ground station measurements of PM2.5 and PM10. PM2.5 and PM10 are highly correlated¹⁰ (Colmer 2020), so one can reasonably be used to estimate the presence of the other. For this reason, while I primarily study impacts of CAAA PM10 regulations, my main outcome of interest is mean annual PM2.5 (μ g/m³) at the tract level. My non-attainment data comes from the EPA Green Book.

6 Descriptive Statistics

6.1 Variable Description

Table 1(a) examines average, tract-level statistics for my variables of interest in 1992. I split the data based on whether their tracts were in attainment of the 1990 EPA PM10 standards. 1992 is significant because it is the year that I specify 90th percentile P.O.C. tracts (1990 Census), and the first year of EPA's PM enforcement. Table 1(b) maintains 1990 Census data while including mean PM concentrations for all years in my sample and splits the data based on whether the tracts were ever in non-attainment.

High (90th percentile) minority tracts comprise around 8% more of total non-attainment tracts than attainment tracts. This relatively small figure confirms that the EJ disparity is driven by high minority tracts being on

¹⁰ Colmer et al. notes that "prior to the year 2000, data on PM10 were used in place of ground station and satellite data on PM2.5, when PM2.5 records were unavailable. All estimates indicate that there is very high persistence in rank over time, irrespective of which base year is used." Many studies have examined this correlation, with it ranging from R=0.64 (Munir et al. 2016) to R=0.95 (Janssen et al. 2013). For my purposes, this evidence is sufficient.
the upper tail of PM distribution rather than by a significantly higher number of communities of color being in non-attainment. Furthermore, across both splits, income is notably higher in non-attainment tracts. Compared to the data on racial groups in non-attainment, income seems relatively insignificant as a predictor of PM.

) when r who is chartannien	111992 - 0	when Philo Nonattainment in 1992 -				
N = 53,969 tracts/observat	N = 10,728 tracts/observations					
Variable	Mean	Mean	t-test: Mean(0) - Mean(1)			
Mean PM2.5 µg/m3	15.08	17.27	-2.190***			
Population	3,825.58	3,773.41	52.169***			
Proportion 90th Percentile Minority Tracts	0.09	0.17	-0.088***			
Share Population Minority	0.19	0.31	-0.116***			
Share Population Black	0.12	0.13	0.00			
Share Population Hispanic	0.07	0.18	-0.115***			
Share Population in Poverty	0.13	0.14	-0.009***			
Income per Capita	14,145.02	15,059.87	-914.849***			
(b) When PM10 or PM2 Nonattainment = 0 (1981-2	When PM10 or PM2.5 Nonattainment = 1 (1981-2016)					
N = 37,455 tracts/observat	ions	N = 27,250 tracts/observations				
Mean PM2.5 µg/m3	12.43	15.66	-3.239***			
Population	3,783.47	3,862.55	-79.087***			
Proportion 90th Percentile Minority Tracts	0.07	0.15	-0.082***			
Share Population Minority	0.17	0.26	-0.090***			
Share Population Black	0.11 0.14		-0.034***			
	0.06	0.12	-0.056***			
Share Population Hispanic	0.06	0.12				
Share Population Hispanic Share Population in Poverty	0.08	0.12	0.021***			

 Table 1: Variable Description

 (a) When PM10 Nonattainment in 1992 = 0
 When PM10 Nonattainment in 1992 = 1

6.2 Nationwide net pollution changes

Figure 2 shows key trends that provide a basis for my hypothesis. They display mean PM2.5 concentrations from 1981 through 2016. Subfigure (a) displays concentrations in high-minority tracts against all tracts, where I observe a perceptible shrinking of the racial pollution exposure gap from roughly 4.5 μ g/m³ to 1.5 μ g/m^{3.11} It supports the key finding of Currie et al. (2020) over an expanded time frame. This is a highly

¹¹ One may reasonably observe that while the level change of the gap is negative, the gap in percentage terms of overall pollution is has not significantly changed. Many examine inequality in percentage terms. But for pollution, negative health impacts occur at high concentrations, so I define my EJ gap in level terms to reflect low marginal impact at low concentrations.

important finding, because Schwartz et al. (2021) finds an increase in life expectancy of 0.29 years when a population was exposed to 7 μ g/m³ versus 12 μ g/m³ of PM2.5. This is a drop similar in means to my EJ gap finding, so a gap closure is likely improving disparities in health and life expectancy outcomes. Something must be driving the gap closure, whether policy, demographic change, or an omitted variable. Subfigure (b) motivates my model specification. Pollution trends of high and low minority tracts cluster together independent of poverty levels. Moreover, the heterogeneity based on poverty status disappeared around 1998. These two observations tell us that, while everyone experienced cleaner air over time, people of color experienced environmental inequality and exposure changes in the past 40 years. As such, I leave the discussion here to solely focus on race.



6.3 Maps

Figure 3 maps pollution and attainment status across the country in 1992, the first year of enforcement of the first nationwide Particulate Matter standards (PM10), and in 2005, the first year of the PM2.5 standards. PM2.5 concentrations in 1992 were highest in industrialized areas like the Rust Belt, the Southeast, and Southern California (Figure 3a). Those spatial patterns held, but overall concentrations fell over time (Figure 3b). Non-attainment status with PM10 standards, on the other hand, was primarily concentrated on the West Coast (Figure 3c). Counties in non-attainment for PM2.5 were concentrated in the Rust Belt, the Tri-State Area, and Southern California in 2005 (Figure 3d). There was a disparity between pollution concentration patterns and PM10 non-attainment status in 1992 due to the different particle composition. Despite this, examining the concentrations in 2005 relative to the PM2.5 non-attainment map, it appears the PM10 regulations were successful in reducing pollution.



Figure 4 examines the geographic dispersion of my key demographic variable of interest, high-minority tracts. The maps show the percentage of high-minority tracts within each county. Communities of color are densely concentrated around each coast, with other clusters in the Great Lakes region, the inland southwest, and the inland southeast. Cross-ref

erencing with the PM maps, correlations between race and pollution are highest in the Southeast and Southern California. Figure 4(b) overlays the location of high-minority tracts with non-attainment status. I observe more instances of treatment for communities of color in the Southwest and the Northeast. Figures 4(a) and 4(b) shows that high-minority areas were covered by both the PM10 and PM2.5 rounds of the CAAA regulation.





7 Findings

7.1 Model

I use difference-in-difference (DiD), stacked DiD, and event study models to study the impact of a county being designated as non-attainment on PM2.5 pollution. I estimate my DiD model using two-way fixed effects, evaluating the average treatment effect of county non-attainment status T_c on PM2.5 concentrations in tract i in county c and year t (Y_{ict}) after the county was designated as being in non-attainment (Post_{ct}). I use tract and year-level fixed effects to account for unobserved geography-specific and time-specific confounders. In each regression, all standard errors are clustered at the county level, which allows for tracts located in a single county to have correlated errors. My regression equation is:

$$Y_{ict} = \beta_0 + \beta_1 (Post_{ct} * T_c) + \gamma_i + \gamma_t + \varepsilon_{ict}$$
(1)

Recent literature raises concerns with the TWFE model in staggered treatment contexts (Borusyak and Jaravel 2016). Attainment status differs over time for some counties, with counties entering non-attainment status as early as 1992 and as late as 2009. (Goodman-Bacon 2021). This is likely the case in my study since we would expect heterogeneity in treatment effects due to technological advancements, updated legal code, and the anticipation effect of impending regulatory approval and enforcement (Malani and Reif 2015). Further bias may result from more recently treated counties serving as controls for earlier treated counties.

To address these concerns, I first use an event study, exploiting within county variation in treatment status in "event time," or time since each treated county was placed into non-attainment status.¹² This approach has a further advantage since the visualization of the event study regression allows me to investigate parallel trends in the years leading up to treatment. While maintaining the TWFE, I study heterogeneous policy impacts through the event-year indicator j and collinear time indicators $[t=\tau]$. I estimate the equation:

$$Y_{ict} = \sum_{j=-\tau} \beta_j \mathbf{1}[t=j] + \gamma_i + \gamma_t + \varepsilon_{ict}$$
(2)

¹² For counties that enter, exit, then re-enter non-attainment, the treatment is measured at the first occasion.

After the basic event study, I then interact all event study terms with indicators for high-P.O.C. tracts in a regression to evaluate heterogeneity.

I also implement a 'stacked' difference-in-differences model. This model addresses concerns of time variation in controls by creating 'clean controls' in an event-specific panel dataset (Cengiz et al. 2019; Beatty et al. 2021). For each year where treatment occurs, a dataset is created with 10 years pre- and post-treatment of event time and never-treated counties as the controls. All those datasets are merged ('stacked') to create a dataset where previously treated tracts do not serve as controls for tracts entering non-attainment years in the future. The regression equation is identical to equation (1) with the alternatively constructed data.

I test for differential outcomes from treatment for EJ communities using a triple Difference-in-Differences (DDD) model (Olden and Møen 2020; Gruber 1994; Cunningham 2021). In this case, the initial treatment split is on attainment status (county level), while the additional split is high-minority tracts H_i versus non-high-minority tracts. The outcome Y_{ict} of the DDD model, specified below, denotes PM2.5 levels by year t, tract i, and county c.

 $Y_{ict} = \beta_0 + \beta_1 H_i^* Post_t + \beta_2 T_c^* Post_t + \beta_3 T_c^* H_i^* Post_t + \gamma_i + \gamma_t + \varepsilon_{ict}$ (3)

The parameter of interest is β 3, the coefficient estimating the difference in pollution for high-minority tracts in non-attainment after the policy.

7.2 Regression Results

Table 2 presents my regression results for the impact of attainment status on average PM2.5 concentrations across all counties. Both columns present model results adjusting for tract and year TWFEs, while column 2 implements the stacked regression dataset. Across both specifications, I confirm prior research and find that the CAAA were responsible for the lower pollution in non-attainment tracts relative to never-treated tracts. This result indicates that non-attainment status can explain around 10% of the ~16 μ g/m³ drop in nationwide mean PM2.5 concentrations (Figure 1).

	Table 2					
Difference-in-Difference Results						
	(1)	(2)				
	TWFE DiD	Stacked DiD				
	Mean PM2.5 µg/m3					
Nonattainment (Treatment)	-2.025***	-1.599***				
	(0.237)	(0.220)				
Constant	14.19***	12.60***				
	(0.0473)	(0.00615)				
R-squared	0.939	0.935				
Robust std. errors in parentheses	td. errors in parentheses *** p<0.01, ** p<0.05, * p<0.1					

. .

Figure 5 presents my event-study model results. In the 'before' period, pollution does not appear to follow any identifiable trend, supporting my research design though the parallel trends assumption. I observe an immediate drop in pollution though year four, then again after year ten. This indicates the presence of a composition effect, where all regulatory impacts are reflected in the short term, while older PM10 regulations are driving further decreases in the later event time. I decompose these year-over-year trends in greater detail in the Appendix (Figure 8).





stacked regression models. The DDD controls for the heterogeneity of non-attainment status and high-minority status. The key interaction term in column 2 highlights that high-minority tracts saw a -0.89 μ g/m³ greater drop in PM2.5 on average than non-high-minority tracts after entering non-attainment status. This indicates that non-attainment status can explain around 25% of the 3.5 μ g/m³ decrease in the EJ gap (Figure 2a) over the last 35 years.

	Table 3			
Triple Difference Regression Results				
	(1)	(2)		
	TWFE	Stacked		
	DDD	DDD		
	Mea	n PM2.5 μg/m3		
Nonattainment (Treatment)	-1.815***	-1.464***		
	(0.212)	(0.198)		
High-P.O.C. Treatment Interaction	-1.383***	-0.889***		
	(0.286)	(0.211)		
Constant	14.19***	12.60***		
	(0.0442)	(0.00582)		
R-squared	0.939	0.936		

Robust std. errors in parentheses **** p<0.01, *** p<0.05, ** p<0.1

Lastly, Figure 6 is an additional event study. Treatment groups are split between high and low minority tracts, allowing me to examine differential treatment effects by race (control groups are excluded). In the 'before' period, the change in racial pollution gap does not appear statistically different as evidenced by clustering around zero and overlapping Confidence Interval (C.I.) lines.¹³ Parallel trends are again confirmed, this time for the DDD heterogeneity analysis. Following year zero of event time, it becomes clear that high-minority tracts experienced a greater decrease in PM2.5 than non-high-minority tracts. The policy effect appears to grow stronger over time, with the gap of pollution change expanding rapidly around the 10th year of event time. Again, it appears a composition effect is present, as a majority of the gap closure is likely attributable to PM10 regulations when PM2.5 non-attainment disappears from the data.

¹³ In each regression, all errors are clustered at the county level which allows for tracts located in a single county have correlated errors.



Under all specifications and coefficients, the results indicate that the CAAA were successful both in their stated goal of cutting air pollution and the EJ objective of narrowing the racial pollution gap. As the models increase in complexity, the coefficients shrink, but maintain significance, indicating the fixed effects and stacked treatment groups are capturing unobserved variance while leaving a meaningful signal. This result allows me to reject the null hypothesis that the CAAA did not reduce the racial pollution gap and provides casual evidence that the CAAA were responsible for the change.

7.3 Discussion and Implications

These results and the validity of the model are conditional on a few factors. First, endogeneity bias is a risk in any observational study, so the causality of my results is conditional on non-attainment status being exogenous. While the pre-trends of the event studies support this, it is impossible to prove. Furthermore, as previously discussed in Section 7.1, this analysis faces heterogeneity in treatment effects due to some counties being treated much later than the initial treatment round. As such, it is possible that the early rounds of SIPs and non-attainment designations induced second- or third-hand behavior impacts or policy changes. The event study results indicate that non-attainment status resulted in near-immediate pollution cuts, but it is still possible that an omitted variable responsible for some of the signal, or non-attainment is associated with confounders.

The findings of Currie et al. and the confirmation and expanded validity

of this study provide macro-level good news for advocates of environmental quality and EJ. The concerns of pollution permit trading markets resulting in higher inequality should be further allayed. In taking a federalist approach to air, it appears that SIPs were broadly effective in cutting pollution and smoothing out disparities. Despite being repeatedly plagued by legal battles and political uncertainty, the Clean Air Act is a success story.

8 Heterogeneity Analysis

Decomposing these regression results allows me to examine the heterogeneous impacts of the CAAA. I re-run my DiD for each county placed under non-attainment status, comparing average PM2.5 concentrations over time to all non-attainment counties to obtain county-specific treatment effects. Figure 7 maps the results. Results vary from -7 μ g/m³ in the South Coast Air Basin¹⁴ and the East Coast to an increase of 6 μ g/m³ in other areas of the Southwest. Areas that saw increases in pollution are a minority of the county treatment effects but demonstrate that SIPs do not universally decrease pollution.



Figure 7

¹⁴ Comprised of Los Angeles and Orange County, California

Table 4 presents correlations between my county-level treatment effects and county demographics (Table 5 in the appendix shows all correlations). Higher PM2.5 reductions correlate with a higher Black population and negatively correlate with the Hispanic population. The latter is a surprising finding given the regression results but makes sense if we compare Figure 4(b) to Figure 7. The treated counties with the greatest increases in pollution overlap some areas with the greatest concentrations of Hispanic population, such as Arizona¹⁵, New Mexico, and Imperial County, California. The negative coefficients on income per capita and percentage population with a high school degree indicate that, consistent with my discussion in Section 4, wealthier and better-educated counties saw more pollution reductions on average.

variable	correlation with beta					
% White	-0.0126					
% Black	-0.391					
% Hispanic	0.3296					
% < 18y/o	0.2455					
% > 60 y/o	-0.0149					
% P.O.C.	-0.0404					
income	-0.097					
% High School Degree	-0.2659					
% manufacturing	-0.313					
% poverty	0.0967					

le 4

Another interesting result is the positive correlation between high PM reductions and county share of manufacturing employment. Using the same variable, Colmer et al. (2020) found that lower manufacturing shares led to higher tract percentile rank points of PM2.5 from 1981 to 2016. These two results suggest that PM2.5 SIPs effectively regulate direct source pollution and potentially less effective in reducing non-point source pollution. Direct sources such as power plants, smokestacks, and construction sites are less mobile and easier to regulate. Non-point sources include wildfires and automobiles, whose emissions interact with NO2, SO2, and VOCs¹⁶ in the atmosphere to form PM2.5. Using Arizona as a case study, the state had multiple counties in non-attainment

¹⁵ Cochise, Pinal, Gila, and Santa Cruz Counties in Arizona saw 4 out of the 6 greatest PM2.5 increases in my treated counties dataset.

¹⁶ See section 2.2

for years and saw PM2.5 increases over time. Primary sources of PM2.5 in Arizona include automobiles and cross-border pollution spillovers from Mexico (ADEQ 2022). Located in a valley, Phoenix, AZ has the 7th worst air in the country, with PM2.5 building up from auto traffic, wood fire pits, and fireworks (Stone 2020). So, with manufacturing not a major source of PM2.5 in Arizona, it seems that the state's SIPs have been ineffective in tackling air pollution problems, which manifests as an EJ disparity given the state's significant Hispanic population.

Conversely, there are significant Black populations in the major metropolitan areas of formerly industrial states like Pennsylvania, California, Ohio, Tennessee, and Georgia. These are the states with counties that saw the greatest pollution drops. Thus, the SIPs targeting manufacturing likely had a strong clean-up effect in these areas, which drove the closure of the EJ gap. However, it is important to remember that the CAAA is only statistically attributable to a portion of pollution reductions (as evidenced in this paper and Currie et al. 2020). Deindustrialization was well underway during the CAAA and was accelerated in the 1990s by the signing of NAFTA. Furthermore, as explained in Section 4.3, the Porter Hypothesis leads me to believe that high-POC areas experienced disproportionate gains in clean industrial technology, reducing pollution. Clearly, there were multiple mechanisms at play that make it difficult to pinpoint a counterfactual. These findings are not causal but demonstrate that policy had heterogeneous impacts across different communities.

9 Conclusion

My research robustly finds a decrease in the racial pollution gap and causal evidence that the Clean Air Act was responsible. In summary, I contribute to the literature on several fronts: I add to the burgeoning utilization of satellite PM data in the Meng, Colmer, and Currie papers. I attempt to replicate and extend the results of Currie et al. (2020) who found CAAA's causal reduction in the Black-White gap of PM2.5. I extend the analysis to include PM10 attainment's impact on PM2.5, which brings in 11 years of extra observations. Event study results suggest that a composition effect is present in the results, with PM10 significantly contributing to the EJ gap closure in the long run. I also extend the analysis to Hispanic communities, confirming similar statistical trends and causal identification. I do so by situating this research within the ongoing empirical debate about observational design using Difference-in-Difference. To

prevent bias from heterogeneous treatment effects, I confirm my results using a diverse range of specifications: traditional Difference-in-Difference, Triple Difference, Event Study, Two Way Fixed Effects, and Stacked Regression. My research robustly finds a decrease in the racial pollution gap and causal evidence that the Clean Air Act Amendments were responsible. Lastly, I conduct a heterogeneity analysis, decomposing earlier results and showing that air pollution decreases in heavy manufacturing counties were a primary driver of the EJ gap closure over time, while stagnating or worsening air quality in predominately Hispanic communities remains.

Disparate contributions of Black versus Hispanic populations to the EJ gap and its closures is worth future consideration. The results of Currie et al. and this paper suggest that gains in Black communities are driving the closure, while Hispanic communities have seen inconsistent outcomes. Future research should examine this question through decomposition and investigate the roles of residential sorting and PM pollution source types (point versus mobile) on these outcomes. It would also be valuable to extend the heterogeneity to more groups contained within regulators' definition of EJ groups, such as Asian and Indigenous populations. Additional next steps in the research should address the dearth of literature examining SIPs. While specific county/regional policies have been analyzed, I have not found a paper that comprehensively summarizes different regulatory structures of SIPs. Such research would allow comparative examination of effective plans to cut pollution, reduce EJ disparities, or improve public health.

Appendix





(b) Event Study for Tracts treated in 2005



Here I decompose the event study regression into tracts that entered non-attainment in 1992 and 2005, the first years of the PM10 and PM2.5 policies. In the former, I observe random distributions prior to the policy, then a noisy but discernable signal of the treatment effect. The original set of SIPs appears to have an immediate, strong treatment effect that is consistent over time. This is supported by Figure 5, the primary event study of all treated tracts, where the signal increases in strength after 10 years of event time (the only tracts remaining after 11 years of event time are those from PM10 regulations). This is likely due to a composition effect of PM2.5 and 10 policies which disappears once PM2.5 data runs out. In the 2005 round, the pre-trends appear to show that the PM10 regulations were impacting the areas soon to enter PM2.5 non-attainment. There is still a significant post-policy treatment effect that is smaller than PM10 initially but grows over time.

variable correlations	beta	% White	% Black	% Hispani c	% < 18 y/o	% > 60 y/o	% P.O.C.	incom c	% GE D	% manu-fa cturing	% povert y
beta	1.000										
% White	-0.013	1.000									
% Black	-0.391	-0.626	1.000								
% Hispanic	0.330	-0.718	-0.04 4	1.000							
% < 18 y/o	0.246	-0.085	-0.12 4	0.270	1.000						
% > 60 y/o	-0.015	0.177	-0.09 1	-0.145	-0.506	1.000					
% P.O.C.	-0.040	-0.989	0.685	0.697	0.108	-0.171	1.000				
income	-0.097	-0.067	0.064	-0.070	-0.363	-0.272	-0.005	1.000			
% GED	-0.266	0.189	0.026	-0.168	-0.122	0.597	-0.104	-0.656	1.000		
% manufacturing	-0.313	0.381	-0.10 6	-0.330	0.082	0.087	-0.316	-0.227	0.413	1.000	
% poverty	0.097	-0.391	0.241	0.336	0.008	0.377	0.418	-0.612	0.450	-0.195	1.000

- 11	r 1	1 1		
	12	n	P	5
	ιu	U	LC.	\sim

Work Cited

- Aaronson, Daniel, Daniel Hartley, and Bhashkar Mazumder. 2021. "The Effects of the 1930s HOLC 'Redlining' Maps." American Economic Journal: Economic Policy 13 (4): 355–92. https://doi.org/10.1257/ pol.20190414.
- ADEQ. 2022. "Timeline of Particulate Matter (PM) National Ambient Air Quality Standards (NAAQS) | US EPA." Arizona Department of Environmental Quality. March 8, 2022. https://www.epa.gov/ pm-pollution/timeline-particulate-matter-pm-national-ambientair-quality-standards-naaqs.
- Aldy, Joseph E., Maximilian Auffhammer, Maureen Cropper, Arthur Fraas, and Richard Morgenstern. 2022. "Looking Back at 50 Years of the Clean Air Act." Journal of Economic Literature 60 (1): 179– 232. https://doi.org/10.1257/jel.20201626.
- Ashman, Hero, and Seth Neumuller. 2020. "Can Income Differences Explain the Racial Wealth Gap? A Quantitative Analysis." Review of Economic Dynamics 35 (January): 220–39. https://doi. org/10.1016/j.red.2019.06.004.
- Atkinson, Anthony B. 1999. "The Contributions of Amartya Sen to Welfare Economics." Scandinavian Journal of Economics 101 (2): 173–90. https://doi.org/10.1111/1467-9442.00151.
- Auffhammer, Maximilian, Antonio M. Bento, and Scott E. Lowe. 2009. "Measuring the Effects of the Clean Air Act Amendments on Ambient PM10 Concentrations: The Critical Importance of a Spatially Disaggregated Analysis." Journal of Environmental Economics and Management 58 (1): 15–26. https://doi. org/10.1016/j.jeem.2008.12.004.
- Banzhaf, H. Spencer, and Randall P. Walsh. 2013. "Segregation and Tiebout Sorting: The Link between Place-Based Investments and Neighborhood Tipping." Journal of Urban Economics 74 (March): 83–98. https://doi.org/10.1016/j.jue.2012.09.006.
- Banzhaf, Spencer, Lala Ma, and Christopher Timmins. 2019. "Environmental Justice: The Economics of Race, Place, and Pollution." Journal of Economic Perspectives 33 (1): 185–208. https://doi.org/10.1257/jep.33.1.185.
- Beatty, Timothy K. M., Gabriel E. Lade, and Jay Shimshack. 2021. "Hurricanes and Gasoline Price Gouging." Journal of the Association of Environmental and Resource Economists 8 (2): 347–74. https:// doi.org/10.1086/712419.
- Bell, Michelle L., Kathleen Belanger, Keita Ebisu, Janneane F. Gent,

- Hyung Joo Lee, Petros Koutrakis, and Brian P. Leaderer. 2010. "Prenatal Exposure to Fine Particulate Matter and Birth Weight: Variations by Particulate Constituents and Sources." Epidemiology 21 (6): 884–91. https://doi.org/10.1097/EDE.0b013e3181f2f405.
- Borusyak, Kirill, and Xavier Jaravel. 2016. "Revisiting Event Study Designs." SSRN Electronic Journal. https://doi.org/10.2139/ ssrn.2826228.
- Boustan, Leah Platt. 2013. "Racial Residential Segregation in American Cities." National Bureau of Economic Research, no. w19045 (May). https://doi.org/10.3386/w19045.
- Bullard, Robert D., Paul Mohai, Robin Saha, and Beverly Wright. 2008. "TOXIC WASTES AND RACE AT TWENTY: WHY RACE STILL MATTERS AFTER ALL OF THESE YEARS." Environmental Law 38 (2): 371–411.
- Callan, Scott J., and Janet M. Thomas. n.d. Environmental Economics & Management: Theory, Policy and Applications. 6th ed. Cengage Learning.
- Cengiz, Doruk, Arindrajit Dube, Attila Lindner, and Ben Zipperer. 2019. "The Effect of Minimum Wages on Low-Wage Jobs*." The Quarterly Journal of Economics 134 (3): 1405–54. https://doi.org/10.1093/ qje/qjz014.
- Chakrabarti, Snigdha, and Nita Mitra. 2005. "Economic and Environmental Impacts of Pollution Control Regulation on Small Industries: A Case Study." Ecological Economics 54 (1): 53–66. https://doi.org/10.1016/j.ecolecon.2004.09.019.
- Cicatiello, Lorenzo, Salvatore Ercolano, and Giuseppe Lucio Gaeta. 2015. "Income Distribution and Political Participation: A Multilevel Analysis." Empirica 42 (2): 447–79. https://doi.org/10.1007/s10663-015-9292-4.
- Coase, R. H. 1960. "The Problem of Social Cost." The Journal of Law & Economics 3: 1–44.
- Colmer, Jonathan. 2020. "Replication Data for: Disparities in PM2.5 Air Pollution in the United States." University of Virginia Dataverse. https://doi.org/10.18130/V3/QLBNQJ.
- Colmer, Jonathan, Ian Hardman, Jay Shimshack, and John Voorheis. 2020. "Disparities in PM2.5 Air Pollution in the United States." Science 369 (6503): 575–78. https://doi.org/10.1126/science.aaz9353.
- Commission for Racial Justice. 1987. "Toxic Wastes and Race in the United States. A National Report on the Racial and Socio-Economic Characteristics of Communities with Hazardous Waste Sites. United Church of Christ." United Church of Christ. https:// www.nrc.gov/docs/ML1310/ML13109A339.pdf.

- Cunningham, Scott. 2021. Causal Inference: The Mixtape. New Haven ; London: Yale University Press.
- Currie, Janet, John Voorheis, and Reed Walker. 2020. "What Caused Racial Disparities in Particulate Exposure to Fall? New Evidence from the Clean Air Act and Satellite-Based Measures of Air Quality." National Bureau of Economic Research w26659 (January). https:// doi.org/10.3386/w26659.
- Cushing, Lara, Dan Blaustein-Rejto, Madeline Wander, Manuel Pastor, James Sadd, Allen Zhu, and Rachel Morello-Frosch. 2018. "Carbon Trading, Co-Pollutants, and Environmental Equity: Evidence from California's Cap-and-Trade Program (2011–2015)." Edited by Jonathan Patz. PLOS Medicine 15 (7): e1002604. https://doi. org/10.1371/journal.pmed.1002604.
- Cushing, Lara J, Madeline Wander, Rachel Morello-Frosch, Manuel Pastor, Allen Zhu, and James Sadd. 2016. "A Preliminary Environmental Equity Assessment Of California's Cap-and-Trade Program." PERE Publications. USC Dana and David Dornsife College of Letters, Arts and Sciences: PERE at USC. http://dornsifelive.usc.edu/PERE/ enviro-equity-CA-cap-trade.
- Darity, William, Darrick Hamilton, Mark Paul, Alan Aja, Anne Price, Antonio Moore, and Caterina Chiopris. 2018. "What We Get Wrong About Closing The Racial Wealth Gap." Samuel DuBois Cook Center on Social Equity. https://socialequity.duke.edu/wpcontent/uploads/2019/10/what-we-get-wrong.pdf.
- Dechezleprêtre, Antoine, and Misato Sato. 2017. "The Impacts of Environmental Regulations on Competitiveness." Review of Environmental Economics and Policy 11 (2): 183–206. https://doi. org/10.1093/reep/rex013.
- Deryugina, Tatyana, Nolan Miller, David Molitor, and Julian Reif. 2020. "Geographic and Socioeconomic Heterogeneity in the Benefits of Reducing Air Pollution in the United States." National Bureau of Economic Research w27357 (June). https://doi.org/10.3386/ w27357.
- Evans, Gary W., and Elyse Kantrowitz. 2002. "Socioeconomic Status and Health: The Potential Role of Environmental Risk Exposure." Annual Review of Public Health 23 (1): 303–31. https://doi. org/10.1146/annurev.publhealth.23.112001.112349.
- Fowlie, Meredith, Stephen P Holland, and Erin T Mansur. 2012. "What Do Emissions Markets Deliver and to Whom? Evidence from Southern California's NOx Trading Program." American Economic Review 102 (2): 965–93. https://doi.org/10.1257/aer.102.2.965.

Garcia, Cynthia A., Poh-Sin Yap, Hye-Youn Park, and Barbara L. Weller.

- 2016. "Association of Long-Term PM2.5 Exposure with Mortality Using Different Air Pollution Exposure Models: Impacts in Rural and Urban California." International Journal of Environmental Health Research 26 (2): 145–57. https://doi.org/10.1080/09603123.2015.1 061113.
- Goodman-Bacon, Andrew. 2021. "Difference-in-Differences with Variation in Treatment Timing." Journal of Econometrics 225 (2): 254–77. https://doi.org/10.1016/j.jeconom.2021.03.014.
- Gray, Wayne B., Ronald J. Shabegian, Ann Wolverton, Wayne B. Gray, Ronald J. Shabegian, and Ann Wolverton. 2010. "Environmental Justice: Do Poor and Minority Populations Face More Hazards?" https://doi.org/10.22004/AG.ECON.280891.
- Greenstone, Michael. 2002. "The Impacts of Environmental Regulations on Industrial Activity: Evidence from the 1970 and 1977 Clean Air Act Amendments and the Census of Manufactures." Journal of Political Economy 110 (6): 1175–1219. https://doi. org/10.1086/342808.
- Greenstone, Michael. 2004. "Did the Clean Air Act Cause the Remarkable Decline in Sulfur Dioxide Concentrations?" Journal of Environmental Economics and Management 47 (3): 585–611. https://doi.org/10.1016/j.jeem.2003.12.001.
- Gruber, Jonathan. 1994. "The Incidence of Mandated Maternity Benefits." The American Economic Review 84 (3): 622–41.
- Hailu, Asnake, and Wasserman Cathy. 2016. "Guidelines for Using Rural-Urban Classification Systems for Community Health Assessment." Washington State Department of Health,.
- Hamilton, James T. 1995. "Testing for Environmental Racism: Prejudice, Profits, Political Power?" Journal of Policy Analysis and Management 14 (1): 107–32. https://doi.org/10.2307/3325435.
- Hausman, Catherine, and Samuel Stolper. 2020. "Inequality, Information Failures, and Air Pollution." National Bureau of Economic Research w26682 (January). https://doi.org/10.3386/w26682.
- Hernandez-Cortes, Danae, and Kyle Meng. 2021. "Do Environmental Markets Cause Environmental Injustice? Evidence from California's Carbon Market." National Bureau of Economic Research, April. https://doi.org/10.3386/w27205.
- House of Representatives, Congress. 2017. 42 U.S.C. 7401 Congressional Findings and Declaration of Purpose. United States Code. https:// www.govinfo.gov/app/details/USCODE-2017-title42/USCODE-2017-title42-chap85-subchapI-partA-sec7401.

- Hsiang, Solomon, Paulina Oliva, and Reed Walker. 2019. "The Distribution of Environmental Damages." Review of Environmental Economics and Policy 13 (1): 83–103. https://doi.org/10.1093/reep/rey024.
- Ito, Koichiro, and Shuang Zhang. 2020. "Willingness to Pay for Clean Air: Evidence from Air Purifier Markets in China." Journal of Political Economy 128 (5): 1627–72. https://doi.org/10.1086/705554.
- Janssen, Nicole, Paul Fischer, Marten Marra, Caroline Ameling, and Flemming Cassee. 2013. "Short-Term Effects of PM2.5, PM10 and PM2.5-10 on Daily Mortality in The Netherlands." The Science of the Total Environment 463-464C (June): 20–26. https://doi. org/10.1016/j.scitotenv.2013.05.062.
- Kim, Sun-Young, Casey Olives, Lianne Sheppard, Paul D. Sampson, Timothy V. Larson, Joshua P. Keller, and Joel D. Kaufman. 2017.
 "Historical Prediction Modeling Approach for Estimating Long-Term Concentrations of PM 2.5 in Cohort Studies before the 1999 Implementation of Widespread Monitoring." Environmental Health Perspectives 125 (1): 38–46. https://doi.org/10.1289/EHP131.
- Kochanek, Kenneth D., Jiaquan Xu, and Elizabeth Arias. 2020. "Mortality in the United States, 2019." NCHS Data Brief, no. 395 (December): 1–8.
- Konisky, David M. 2009. "The Limited Effects of Federal Environmental Justice Policy on State Enforcement." Policy Studies Journal 37 (3): 475–96. https://doi.org/10.1111/j.1541-0072.2009.00324.x.
- Leeves, Gareth D., and Ric D. Herbert. 2002. "Economic and Environmental Impacts of Pollution Control in a System of Environment and Economic Interdependence." Chaos, Solitons & Fractals 13 (4): 693–700. https://doi.org/10.1016/S0960-0779(01)00003-0.
- Levinson, Arik, and M. Scott Taylor. 2008. "Unmasking the Pollution Haven Effect." International Economic Review 49 (1): 223–54. https://doi.org/10.1111/j.1468-2354.2008.00478.x.
- Logan, Trevon D., and John M. Parman. 2017. "The National Rise in Residential Segregation." The Journal of Economic History 77 (1): 127–70. https://doi.org/10.1017/S0022050717000079.
- Malani, Anup, and Julian Reif. 2015. "Interpreting Pre-Trends as Anticipation: Impact on Estimated Treatment Effects from Tort Reform." Journal of Public Economics 124 (April): 1–17. https:// doi.org/10.1016/j.jpubeco.2015.01.001.
- Mansur, Erin T., and Glenn Sheriff. 2021. "On the Measurement of Environmental Inequality: Ranking Emissions Distributions Generated by Different Policy Instruments." Journal of the Association of Environmental and Resource Economists 8 (4): 721–58. https://doi.org/10.1086/713113.

- McGuire, Martin C. 1982. "Regulation, Factor Rewards, and International Trade." Journal of Public Economics 17 (3): 335–54. https://doi. org/10.1016/0047-2727(82)90069-X.
- McKernan, Signe-Mary, Caroline Ratcliffe, Margaret Simms, and Sisi Zhang. 2014. "Do Racial Disparities in Private Transfers Help Explain the Racial Wealth Gap? New Evidence From Longitudinal Data." Demography 51 (3): 949–74. https://doi.org/10.1007/s13524-014-0296-7.
- McKitrick, Ross. 2007. "Why Did US Air Pollution Decline after 1970?" Empirical Economics 33 (3): 491–513. https://doi.org/10.1007/ s00181-006-0111-4.
- Meng, Jun, Chi Li, Randall V. Martin, Aaron van Donkelaar, Perry Hystad, and Michael Brauer. 2019. "Estimated Long-Term (1981–2016) Concentrations of Ambient Fine Particulate Matter across North America from Chemical Transport Modeling, Satellite Remote Sensing, and Ground-Based Measurements." Environmental Science & Technology 53 (9): 5071–79. https://doi.org/10.1021/acs. est.8b06875.
- Michener, Jamila D. 2017. "People, Places, Power: Medicaid Concentration and Local Political Participation." Journal of Health Politics, Policy and Law 42 (5): 865–900. https://doi.org/10.1215/03616878-3940468.
- Millimet, Daniel L., and John A. List. 2004. "The Case of the Missing Pollution Haven Hypothesis." Journal of Regulatory Economics 26 (3): 239–62. https://doi.org/10.1007/s11149-004-7550-7.
- Mohai, Paul, and Robin Saha. 2007. "Racial Inequality in the Distribution of Hazardous Waste: A National-Level Reassessment." Social Problems 54 (3): 343–70. https://doi.org/10.1525/sp.2007.54.3.343.
- Montgomery, W.David. 1972. "Markets in Licenses and Efficient Pollution Control Programs." Journal of Economic Theory 5 (3): 395–418. https://doi.org/10.1016/0022-0531(72)90049-X.
- Munir, Said, Turki M. Habeebullah, Atef M. F. Mohammed, Essam A. Morsy, Mohammad Rehan, and Kawsar Ali. 2017. "Analysing PM2.5 and Its Association with PM10 and Meteorology in the Arid Climate of Makkah, Saudi Arabia." Aerosol and Air Quality Research 17 (2): 453–64. https://doi.org/10.4209/aaqr.2016.03.0117.
- Olden, Andreas, and Jarle Møen. 2020. "The Triple Difference Estimator." Discussion Papers from Norwegian School of Economics, Department of Business and Management Science 1 (April). https:// econpapers.repec.org/paper/hhsnhhfms/2020_5f001.htm.

- Olson, Mancur. 1965. The Logic of Collective Action: Public Goods and the Theory of Groups. Harvard Economic Studies 124. Cambridge (Mass.) London: Harvard university press.
- Pope, C. Arden. 1991. "Respiratory Hospital Admissions Associated with PM 10 Pollution in Utah, Salt Lake, and Cache Valleys." Archives of Environmental Health: An International Journal 46 (2): 90–97. https://doi.org/10.1080/00039896.1991.9937434.
- Pope, C. Arden, Majid Ezzati, and Douglas W. Dockery. 2009. "Fine-Particulate Air Pollution and Life Expectancy in the United States." New England Journal of Medicine 360 (4): 376–86. https://doi. org/10.1056/NEJMsa0805646.
- Pope III, C. Arden. 2002. "Lung Cancer, Cardiopulmonary Mortality, and Long-Term Exposure to Fine Particulate Air Pollution." JAMA 287 (9): 1132. https://doi.org/10.1001/jama.287.9.1132.
- Reitze, Arnold W. Jr. 2004. "Air Quality Protection Using State Implementation Plans -Thirty-Seven Years of Increasing Complexity." Villanova Environmental Law Journal 15 (2): 209– 366.
- Ringquist, Evan J. 2005. "Assessing Evidence of Environmental Inequities: A Meta-Analysis." Journal of Policy Analysis and Management 24 (2): 223–47. https://doi.org/10.1002/pam.20088.
- Ringquist, Evan J. 2011. "Trading Equity for Efficiency in Environmental Protection? Environmental Justice Effects from the SO2 Allowance Trading Program*: Trading Equity for Efficiency in Environmental Protection?" Social Science Quarterly 92 (2): 297–323. https://doi. org/10.1111/j.1540-6237.2011.00769.x.
- Schwartz, Joel D., Qian Di, Weeberb J. Requia, Francesca Dominici, and Antonella Zanobetti. 2021. "A Direct Estimate of the Impact of PM2.5, NO2, and O3 Exposure on Life Expectancy Using Propensity Scores." Epidemiology (Cambridge, Mass.) 32 (4): 469– 76. https://doi.org/10.1097/EDE.00000000001354.
- Sen, Amartya. 1979. "Equality of What?" Presented at the The Tanner Lecture on Human Values, Stanford University, May 22. https:// www.ophi.org.uk/wp-content/uploads/Sen-1979_Equality-of-What.pdf.
- Sen, Amartya. 1982. Choice, Welfare, and Measurement. 1st MIT Press edition. Cambridge, Mass: MIT Press.
- Shadbegian, Ronald J., and Ann Wolverton. 2010. "Location Decisions of U.S. Polluting Plants: Theory, Empirical Evidence, and Consequences." National Center for Environmental Economics, EPA, NCEE Working Paper Series, May. https://ideas.repec.org/p/ nev/wpaper/wp201005.html.

- Shapiro, Joseph S., and Reed Walker. 2018. "Why Is Pollution from US Manufacturing Declining? The Roles of Environmental Regulation, Productivity, and Trade." American Economic Review 108 (12): 3814–54. https://doi.org/10.1257/aer.20151272.
- Shapiro, Joseph S., and Reed Walker. 2021. "Where Is Pollution Moving? Environmental Markets and Environmental Justice." AEA Papers and Proceedings 111 (May): 410–14. https://doi.org/10.1257/ pandp.20211004.
- Shaw, Al, and Lylla Younes. 2021. "The Most Detailed Map of Cancer-Causing Industrial Air Pollution in the U.S." ProPublica. November 2, 2021. https://projects.propublica.org/toxmap/.
- Shehab, M. A., and F. D. Pope. 2019. "Effects of Short-Term Exposure to Particulate Matter Air Pollution on Cognitive Performance." Scientific Reports 9 (1): 8237. https://doi.org/10.1038/s41598-019-44561-0.
- Stingone, Jeanette A., Katharine H. McVeigh, and Luz Claudio. 2016. "Association between Prenatal Exposure to Ambient Diesel Particulate Matter and Perchloroethylene with Children's 3rd Grade Standardized Test Scores." Environmental Research 148 (July): 144–53. https://doi.org/10.1016/j.envres.2016.03.035.
- Stone, Erin. 2020. "As Arizonans Battle Respiratory Pandemic, Phoenix Air Quality Earns Poor Marks." The Arizona Republic. April 23, 2020. https://www.azcentral.com/story/news/local/arizonaenvironment/2020/04/23/arizona-battles-covid-19-phoenix-airearns-bad-marks/3008574001/.
- Stuart, Amy L., Sarntharm Mudhasakul, and Watanee Sriwatanapongse. 2009. "The Social Distribution of Neighborhood-Scale Air Pollution and Monitoring Protection." Journal of the Air & Waste Management Association 59 (5): 591–602. https://doi. org/10.3155/1047-3289.59.5.591.
- Tessum, Christopher W., David A. Paolella, Sarah E. Chambliss, Joshua S. Apte, Jason D. Hill, and Julian D. Marshall. 2021. "PM2.5 Polluters Disproportionately and Systemically Affect People of Color in the United States." Science Advances 7 (18): eabf4491. https://doi.org/10.1126/sciadv.abf4491.
- US EPA. 2018. Revisions to Appendix P to 40 CFR Part 51, Concerning Minimum Emission Reporting Requirements in SIPs. RIN 2060–AT80. https://mobile.reginfo.gov/public/do/ eAgendaViewRule?pubId=201710&RIN=2060-AT80.
- US EPA. 2020. "Evolution of the Clean Air Act | US EPA." https://www. epa.gov/clean-air-act-overview/evolution-clean-air-act.

- US EPA. 2021a, May 17. "Basic Information about Air Quality SIPs." https://www.epa.gov/sips/basic-information-air-quality-sips.
- US EPA. 2021c, August 16. "Criteria Air Pollutants." https://www.epa. gov/criteria-air-pollutants.
- US EPA. 2021d, May 26. "Health and Environmental Effects of Particulate Matter (PM)." https://www.epa.gov/pm-pollution/health-andenvironmental-effects-particulate-matter-pm.
- US EPA. 2021b, January 19. "Timeline of Particulate Matter (PM) National Ambient Air Quality Standards (NAAQS) | US EPA." https://www.epa.gov/pm-pollution/timeline-particulate-matterpm-national-ambient-air-quality-standards-naaqs.
- USDA ERS. 2020. "Rural-Urban Commuting Area Codes." https://www. ers.usda.gov/data-products/rural-urban-commuting-area-codes. aspx.
- Webb Hooper, Monica, Anna María Nápoles, and Eliseo J. Pérez-Stable. 2020. "COVID-19 and Racial/Ethnic Disparities." JAMA 323 (24): 2466. https://doi.org/10.1001/jama.2020.8598.
- Whipple, Hannah. 2021. "The Effect of Racial Covenants on Modern Day Foreclosures: Evidence from Hennepin County." Macalester College Economics Honors Projects., no. 109. https://digitalcommons. macalester.edu/economics_honors_projects/109/.
- Williams, Robert B. 2017. "Wealth Privilege and the Racial Wealth Gap: A Case Study in Economic Stratification." The Review of Black Political Economy 44 (3-4): 303-25. https://doi.org/10.1007/ s12114-017-9259-8.
- Wu, Rui, Hancheng Dai, Yong Geng, Yang Xie, Toshihiko Masui, Zhiqing Liu, and Yiying Qian. 2017. "Economic Impacts from PM 2.5 Pollution-Related Health Effects: A Case Study in Shanghai." Environmental Science & Technology 51 (9): 5035–42. https://doi. org/10.1021/acs.est.7b00026.
- Wu, X., R. C. Nethery, M. B. Sabath, D. Braun, and F. Dominici. 2020. "Air Pollution and COVID-19 Mortality in the United States: Strengths and Limitations of an Ecological Regression Analysis." Science Advances 6 (45): eabd4049. https://doi.org/10.1126/sciadv. abd4049.
- Zivin, Joshua Graff, and Matthew Neidell. 2012. "The Impact of Pollution on Worker Productivity." American Economic Review 102 (7): 3652–73. https://doi.org/10.1257/aer.102.7.3652.
- Zou, Eric Yongchen. 2021. "Unwatched Pollution: The Effect of Intermittent Monitoring on Air Quality." American Economic Review 111 (7): 2101–26. https://doi.org/10.1257/aer.20181346.



Berkeley Economic Review